Integrating Reuse in MaTrace Models

An implementation and evaluation

Raphael Andreas Elbing





Integrating Reuse in MaTrace Models

An implementation and evaluation

by

Raphael Andreas Elbing

Den Haag, 25th of August 2022

Student Number Universiteit LeidenStudent Number TU Delfts29193035431131

First Supervisor:	Dr. Tomer Fishman, Institute of Environmental Science (CML), Universiteit Leiden
Second Supervisor:	Dr. Benjamin Sprecher, Industrial Design Engineering, Technische Universiteit Delft
Advisor:	Dr. Sebastiaan Deetman, Institute of Environmental Science (CML) Universiteit Leiden





Summary

The growing wold population gaining affluence is driving the extraction of raw materials. Resource availability is finite and concerns about future supply shortages rise. An approach to tackle this problem is circular economy which entails multiple strategies to reduce the demand of virgin materials. The implementation of those strategies require knowledge about material stocks and flows in a society. Material Flow Analysis can provide those insights. This fast developing field brought about MaTrace models which allow to trace the fate of materials in an open-loop recycling system. Recycling is only one of multiple circular strategies, thus the purpose of this research is to integrate an elaborate reuse model into a MaTrace model to build the foundation of a model which considers multiple circular strategies in sufficient ways.

Two existing models were combined to achieve this: Consumer goods present in the MaTrace model were redirected into a reuse model and the end of life products of the reuse model were fed back into the MaTrace model. The impacts of this model extension were investigated by comparing the total in-use stock when considering one, two, and three consumer products' use cycles. Furthermore, Monte Carlo simulations were conducted to gain an understanding of the model behaviour.

The results show that the total in-use stock increases in the peak by 8 % when reuse is considered. However, the gross stock dynamics do not change significantly in comparison to the original model. The evaluation of the Monte Carlo simulations revealed that the input which contributes the most uncertainty to the total in-use stock is the split of the initial material inflow. Furthermore, the results of the Monte Carlo simulations appear to be strongly connected to the initial input data. On the basis of this research it can be recommended to further extend MaTrace models to obtain a more comprehensive representation of a circular economy. Furthermore, MaTrace models using time series data for inflows and model parameter have to be created, this way MaTrace models can follow the evolution of MFA from static to dynamic models.

Contents

Su	ımma	ary		i			
Nomenclature							
Lis	List of Figures						
Lis	st of	Tables		vi			
1	Intro 1.1 1.2 1.3	oductio State (1.1.1 1.1.2 1.1.3 1.1.4 1.1.5 Resea A case	on of the art	1 3 3 4 7 8 9 10 11			
2	Met 2.1 2.2 2.3 2.4	hodolo Model Impac 2.3.1 2.3.2 Monte 2.4.1 2.4.2 2.4.3	bgy elling approach el implementation ct analysis ct analysis Comparison to the MaTrace model created by Godoy León et al. (2020) Impact of reuse e Carlo simulations Model input mapping Framework Evaluation of Monte Carlo Simulations	13 13 15 16 17 17 17 18 19 24			
3	Data 3.1 3.2 3.3 3.4 3.5	a Data a Extend Data c Data c Data c	as present in the models	26 26 26 26 27 28			
4	Res 4.1 4.2 4.3	ults Gener Impac 4.2.1 4.2.2 Monte	eral model output	30 30 32 32 34 37			
5	Disc 5.1 5.2 5.3 5.4 5.5	Model Used o Impac Monte	n eling approach and limitations of the model data and uncertainties ct and model output e Carlo Simulations mmendations	41 41 43 43 44			

6	Conclusion	47
Re	eferences	53
Α	Model input data A.1 Use phase A.2 End-of-life phase A.3 Production	 V
В	Uncertainty scoringB.1Use phaseB.2End-of-life phaseB.3Production	VI VI IX IX
С	Monte Carlo simulations input distributions	XI
D	Model application manual and description of digital appendix	XIII

Nomenclature

Abbreviations

Abbreviation	Definition
CRM	Critical raw material
DQR	Data Quality Rating
DRC	Democratic Republic of Congo
EE	Electronic equipment
EOL	End-of-life
ETM	Energy transition metal
EV	Electric vehicles
GDP	Gross Domestic Product
GHG	Greehnhouse gas
IE	Industrial ecology
IOT	Internet of things
MFA	Material Flow Analysis

List of Figures

1.1 1.2 1.3 1.4	Single dynamic stock	3 5 8 9
2.1 2.2 2.3 2.4 2.5 2.6 2.7 2.8	MaTrace model system diagram by Godoy León et al. (2020)	14 15 17 18 20 22 23 24
4.1 4.2 4.3 4.4 4.5 4.6 4.7 4.8 4.9 4.10	Distribution of stocks and accumulated outflows	30 31 32 33 35 36 37 38 39 39
5.1	Simplified conceptual integration of 9R strategies into the Reuse-MaTrace model	42
C.1 C.2 C.3	Histogram and defined distribution of a transfer coefficient	XI XII XII

List of Tables

1.1 1.2 1.3	9R circular economy framework	2 3 4
2.1 2.2 2.3 2.4 2.5	Cohort matrix example I	16 16 19 19 24
3.1 3.2 3.3 3.4 3.5 3.6	E-Bike transfer coefficients	27 27 28 28 28 29
4.1 4.2 4.3 4.4	Comparison of instant of total in use-stock depletion by models and data sets Comparison of instants of portable batteries in-use stock depletion by models and data sets Comparison of instants of total in-use stock depletion by number of use cycles Comparison of instant of portable batteries in-use stock depletion by number of use cycles	33 34 35 36
A.1 A.2 A.3 A.4 A.5 A.6 A.7 A.8 A.9 A.10 A.11 A.12	Inputs initial flow split and Weibull parameters for all product categories except for portable batteries	
A.13 B.1	Inputs production	V VI
B.2 B.3 B.4 B.5 B.6 B.7 B.8 B.9	Input uncertainty scoring for hibernating stock of all product categories except for portable batteries	VII VII VII VII VIII VIII VIII
B.10	Input uncertainty scoring collection and pre-treatment	IX

B.11 Input uncertainty scoring distribution to recycling processes B and recycling efficiency .							
B.12 Input uncertainty scoring distribution of secondary material and fraction of exported							
secondary material	IX						
B.13 Input uncertainty scoring production	Х						

1

Introduction

Throughout the current century raw material extraction constantly increases. A commonly known driver is a growing, more affluent world population which rises concerns about future resource availability (Calvo et al., 2017). Moreover critical raw materials (CRM) as lithium or cobalt are subject to other drivers. The accelerating energy transition from a fossil fuel based system towards renewable energy sources requires enormous amounts of CRMs for production and storage. Although, there is currently no shortage in supply, the demand due to global climate ambition may exceed possible supply volumes (Pommeret et al., 2022). Additionally, the private use of portable electronics accelerates because of technological developments as internet of things (IOT) or 5G appliances (Desai, 2020) all adding to the demand of CRMs.

These mingled construct of demand and supply holds complex implications regarding sustainability. Looking at the ecological aspect of sustainability one finds that those materials are required to minimize the carbon emissions of the energy system on the one hand. On the other hand, the mining of some materials leads to environmental pollution (Sovacool, 2019). Regarding the social component of sustainability it is apparent that the implications of CRM mining are severe for some materials. Cobalt for example is mainly mined in the Democratic Republic of Congo (DRC) where insecure mining conditions lead to reoccurring human rights violations such as child labor (Sovacool, 2019). For those reasons it is highly relevant for the field of industrial ecology (IE) to study material demand, usage, and supply.

Since the usage of CRMs appears to be unavoidable, losses in production and waste management have to be minimized and resource efficiency has to be optimised. Resource efficiency is broadly defined as the value created by a resource input while also considering environmental impacts. The term resources as it is used in this research covers amongst others raw materials, energy, or water (Hirschnitz-Garbers et al., 2013). In the context of this thesis resources must be understood as the length and intensity of material usage in a system.

An elaborate concept on how to increase resource efficiency is circular economy. The concept entails a model of production and consumption with the target to reduce material waste by moving from a linear economy (take-make-consume-throw away) towards a circular economy ("Circular economy", 2015). Multiple strategies can contribute to the transition into a circular economy. The 9R framework classifies those strategies.

Circular economy		Strategies	Description
Î	Smarter product	R0 Refuse	Make product redundant by abandoning its function or by offering the same function with a radically different product
	use and manufacture	R1 Rethink	Make a product use more intensive (e.g. by sharing product)
		R2 Reduce	Increase efficiency in product manufacture or use by consuming Fewer natural resources and materials
	Extend lifespan of	R3 Reuse	Reuse by another consumer of discarded product which is still in good condition and fulfils its original function
	product and its	R4 Repair	Repair and maintenance of defective product so it can be used with its original function
	parts	R5 Refurbish	Restore an old product and bring it up to date
		R6 Remanufacture	Use parts of discarded product in a new product with the same function
		R7 Repurpose	Use discarded product or its parts in a new product with different function
	Useful	R8 Recycle	Process materials to obtain the same (high grade) or lower (low grade) quality
	of materials	R9 Recover	Incineration of material with energy recovery

Linear

economy

Table 1.1: 9R circular economy framework by Kirchherr et al. (2017). Strategies become less circular from top to bottom.

The Table 1.1 shows, that the framework lists strategies by their circularity. Refuse within the strategy cluster "Smarter product use and manufacture" is claimed to be the most effective one. Reuse makes the fourth place in the ranking within the cluster "Expend lifespan of product and its parts".

But rather than applying those strategies blindly it is important to gain an understanding about society's metabolism first. In this context, the method Material Flow Analysis (MFA) offers important insights since it aims to study and quantify material stocks and flows in defined regions and temporal scopes. The evolution of the method gained momentum in recent decades. MFA started out with static models which consider only the stocks and flows of one year. On this basis, dynamic MFA was developed which allows to model past material flows and stocks based on historical data but also future flows and stocks based on scenarios (Müller et al., 2014). Hence, the tool can offer insights on where to apply which strategy to reach a maximum improvement in resource efficiency.

A model variant which gained popularity over the last years is the MaTrace model. The models specialty is the ability to showcase losses in an open-loop recycling system. It shows the fate of the material by tracking the inflow of one specific year over the coming years (Nakamura et al., 2014, also see Section 1.1.2). But according to the 9R framework recycling is not the most effective strategy to increase resource efficiency. Multiple scholars integrated other strategies as reuse into the original MaTrace model (Klose & Pauliuk, 2021; Pauliuk et al., 2017; Zhang et al., 2021, also see Section 1.1.2). However, the integration or reuse in those models is rather simplistic and does not necessarily match the state of the art in reuse modeling in MFA (see Section 1.1.3).

Therefore, the main target of this master thesis is to integrate reuse in a more elaborate fashion into a MaTrace model and to evaluate its impact and uncertainties. The following chapter of this thesis provides a state of the art about the MaTrace modelling approach, the modelling of reuse, and a description of modeling uncertainties on the example of survival curves in dynamic Material Flow Analysis (MFA). Research gaps and research questions are derived based on this state of the art. This is followed by a brief introduction to the case study. The second chapter provides the methodology describing how existing models were used and how modelling approaches are simplified, followed by a chapter

describing the used data. The results of the research are provided in the fourth chapter, followed by a discussion and a conclusion.

1.1. State of the art

1.1.1. Dynamic MFA and MaTrace models

Primary material production affects economy, society, and environment in various ways. Some metals are scarce (Godoy León et al., 2020) causing dependencies on suppliers who cannot ensure safe working condition and involve child labour as it is the case for cobalt (Sovacool, 2019). Moreover, primary material production leads to GHG emission and other negative environmental impacts (Pauliuk et al., 2017). Because of the economic and environmental relevance, it is important to gain more knowledge about anthropogenic material flows and cycles (Jarrín Jácome et al., 2021).

A commonly used tool which serves this purpose is Material Flow Analysis (MFA). The original model as introduced by Baccini and Brunner (1991) describes material stocks and flows in a region with the temporal scale of one year. Hence, those models represent a snapshot in time without considering any past or future developments. Therefore, one speaks about static MFA when referring to these models.

As described, knowing stock and flow dynamics is important since it provides insights to future resource use, production and waste management planning, and potential environmental problems (B. Müller, 2006). Motivated by these reasons B. Müller (2006) introduced dynamic stock modelling (Sartori et al., 2016) which builds the foundation of dynamic MFA. This method of stock and flow quantification allows for evaluation of model inputs present as time series by calculating the respective stocks and flows. A static MFA model always complies with the law of mass conservation. There are multiple characteristics to differentiate dynamic MFA models. A key characteristic describes whether the model is inflow- or stock-driven. The following figure shows, that there are three elements to be determined; the inflow *I*, the stock *S*, and the outflow *O*.



Figure 1.1: Single dynamic stock S with one inflow I and one outflow O

Stock driven models calculate I and O based on the stock S itself and a survival curve s(t) (see Section 1.1.5 for more explanation on survival curves). In contrast, inflow driven models calculate the S and O based on the inflow I and the survival curve s(t) (Müller et al., 2014). In the following, the calculations for an inflow-driven model as depicted in Figure 1.1 are presented. This concept will be used throughout this thesis. Assuming there is a data set consisting of an input series covering three years and a survival curve (see following table):

Time t and t'	Inflow $I(t)$	Survival curve $s(t')$
0	3	1
1	5	0.8
2	4	0.5

Table 1.2: Example data for inflow driven model

Then the inflow and the survival curve are exogenous model variables since they do not depend on any other variable in this case. The stock and the outflow are endogenous model variables since they depend on the model inputs (Müller et al., 2014). The stock can be calculated using the following formula (inspired by B. Müller, 2006):

$$S(t) = \sum_{\tau=t_0}^{t} I(\tau) \cdot s(t-\tau)$$
(1.1)

The outflow is then calculated via the net flow N:

$$N(t) = S(t) - S(t-1)$$
(1.2)

$$O(t) = I(t) - N(t)$$
 (1.3)

When applied in a model, those calculations are executed via a cohort matrix. The following cohort matrix belongs to the data as presented in Table 1.2:

	0	1	2
0	3	0	0
1	2.4	5	0
2	1.5	4	4

Table 1.3: Example cohort matrix

Each column of the cohort matrix shows the depreciation of the inflow of one year (cohort) over time (rows). Since there is no stock depletion in the first year (see Table 1.2, survival curve s(0) = 1) the diagonal axis holds the inflow. With the help of those matrices the total stock *S* can be calculated by summing up the rows (for more information regarding the basic calculations see Lauinger et al. (2021)).

Besides the modeled system itself, and depending on whether a model is inflow- or stock-driven, models vary in their temporal scope (retrospective or prospective) and their regional scope (e.g., regional, national or global). Furthermore, some models consider hibernating stocks (also see Section 1.1.3).

Although dynamic MFA modelling might appear as straight forward, diverse approaches in extrapolating input data can be found as well. This is done to model the future or to test different scenarios which often rely on drivers. An example for this is the intensity-of-use hypothesis which assumes a relation between the demand of a material and the gross domestic product (GDP) of the region in focus. The review on dynamic MFA of Müller et al. (2014) found that extrapolation methods are often prone to over simplification, for example by not accounting for the scarcity of the material. Hence, interdisciplinarity is required in order to analyse society's metabolism (Lauinger et al., 2021). Dynamic MFA together with dynamic stock modelling forms the basis for MaTrace models. Its purposes and function will be described in the following section.

1.1.2. MaTrace model

Motivation and Purpose

The original MaTrace model of Nakamura et al. (2014) provides an elaborate model to track the fate of materials in an open-loop recycling system. This development is relevant since recycling is a key strategy to reduce the dependency on virgin materials. Therefore, it is essential to understand the basics about it.

Recycling has the potential to bring about closed material cycles requiring only little primary material input. If recycling is in place, the virgin material input depends on the following three factors: The growth rate of the in-use stocks, the ability of waste management facilities to recover a scrap and the ability of recycling companies to produce high quality secondary material while maintaining a high material efficiency (Pauliuk et al., 2017). Besides those factors, recycling can be functional or non functional. Materials recovered in functional recycling are fit to be reused for the same high quality products while materials (Godoy León et al., 2020).

There are, furthermore, factors concerning the actors participating in the use phase which determine the amount of material entering recycling. This depends on the lifetime of products on the one hand (see Section 1.1.3) and on the other hand on the waste separation performance of actors. How well actors separate waste is determined by their knowledge and perception of waste management (Uhunamure

et al., 2021). Minimizing the described losses in recycling is the target of sustainable management and circular economy strategies. To do so, knowledge about the pathway of materials in an open-loop recycling system are required. The original MaTrace model by Nakamura et al. (2014) aims to fill this knowledge gap. It is able to trace the fate of materials which entered the system at beginning of the regarded period. This way the quantity of losses, functional and non functional recycling become apparent (Nakamura et al., 2014). It is important to understand that the model only considers *one cohort*, the virgin material input of only one year. It then traces the whereabouts of this single input over time.

How it works

The precise structure of a MaTrace model varies insignificantly depending on application and scope. Therefore, the fundamental functionalities are explained based on a generic model as displayed in Figure 1.2 clustering the elements into Use, End-of-Life, and Production following Godoy León et al. (2020).



Figure 1.2: Simplified MaTrace system diagram inspired by Nakamura et al. (2014) and Godoy León et al. (2020)

MaTrace models allow to track the whereabouts of a specific material entering the system at the beginning of the regarded period (one cohort, see Figure 1.2; initial products) across products and time. Hence, only one inflow, the initial product distribution, has to be provided. The initial products vector is summed up to 1 (or 100%) since MaTrace models trace the whereabouts of material in percent (Godoy León et al., 2020; Nakamura et al., 2014; Pauliuk et al., 2017; Zhang et al., 2021). The depletion of the initial stock is defined by distributions. Each product category has its own survival curve defined by multiple parameters depending on the distribution. This stock depletion defines the yearly outflow of End-of-Life products (Godoy León et al., 2020; Nakamura et al., 2020; Nakamura et al., 2014).

The processes in the End-of-Life and Production cluster are defined by process yields (efficiencies). Applying the principle of mass conservation, all flows are found. Outflows can also be exports or more specific losses than those presented in Figure 1.2. Furthermore, two allocation matrices were

used. One allocating the shares of the products to different recycling processes and one allocating the secondary material to production processes (here manufacturing and processing) (Godoy León et al., 2020).

Other challenges posed in solving this system are the closed loops between *Manufacturing*, *Scrap*, and *Processing*. Once these loops are mathematically formalized, one finds an infinite geometric series. The following equation shows the series for the total outflow of produced products of the node *Manufacturing* x_k in the year t, where k is the index referring to the different product categories. $p_{Ml,k}$ is the product outflow of one round of recycling where l indexes the round (Godoy León et al., 2020):

$$x_k(t) = p_{M1,k} + p_{M2,k} + p_{M3,k} + \cdots$$
(1.4)

To solve this, an expression for the share of recovered scrap in recycling round one needs to be found first. Here $\xi_{P,k}$ indicates the processing scrap recovery, $\xi_{M,k}$ indicates the manufacturing scrap recovery, $\lambda_{P,k}$ represents the yield of the processing node, and $\lambda_{M,k}$ represents the yield of the manufacturing node (Godoy León et al., 2020):

$$N_k = \xi_{P,k} \cdot (1 - \lambda_{P,k}) + \xi_{M,k} \cdot (1 - \lambda_{M,k}) \cdot \lambda_{P,k}$$

$$(1.5)$$

Using N_k , $\lambda_k = \lambda_{P,k} \cdot \lambda_{M,k}$, and $m_k(t)$ which is the amount of refined materials being available for manufacturing one finds the following infinite geometrical series (Godoy León et al., 2020):

$$x_k(t) = \lambda_k \cdot (1 + N_k + N_k^2 + \dots) \cdot m_k(t)$$

$$(1.6)$$

This series can be solved analytically (Weisstein, 2022) resulting in (Godoy León et al., 2020):

$$x_k(t) = \lambda_k \cdot (1 - N_k)^{-1} \cdot m_k(t)$$
(1.7)

This solution makes the system computable. A full derivation of the expression can be found in either the supporting information of Nakamura et al. (2014) or Godoy León et al. (2020).

Once the whole cycle is computed, the produced goods (equation above) flow back into the use stock and the cycle has to be repeated for each year in the considered time frame (also see Section 2.2). In every iteration the stock depletion of each previous cohort must be considered. This way the fate of the virgin material input of one year is tracked over multiple life- or recycling cycles. Hence, MaTrace does not evaluate the whole material present in the system but only the share which entered the system in one specific year.

Versions

Since the publication of the MaTrace model by Godoy León et al. (2020) of Nakamura et al. (2014), multiple further publications applied or extended the model (Scopus search with search term "MaTrace"). Only Takeyama et al. (2016) left the original model untouched while Godoy León et al. (2020) and Jarrín Jácome et al. (2021) adopted the model to suit the regarded material. Larger changes were implemented by Pauliuk et al. (2017) who developed a global MaTrace model, by Helbig et al. (2022) who traced multiple materials, and by Nakamura et al. (2017) who extended the model to trace the fate of substances in alloys.

Moreover, multiple publications implement circular practices as re-manufacturing (Zhang et al., 2021) and reuse (Klose & Pauliuk, 2021; Pauliuk et al., 2017; Zhang et al., 2019). Pauliuk et al. (2017) treat reuse as a byproduct of the global MaTrace model. A share of the end-of-life products of one region may enter the use phase of another region via trade. An altered lifetime of reused products is not considered.

Klose and Pauliuk (2021) explicitly treat reuse in their model by considering shares of end-of-life products which are reused. However, the considered lifetimes originate from durability standards or product warranty laws. It is explicitly stated that the lifetime of products entering the use phase does not change for reused products and throughout time.

Zhang et al. (2021) investigate the impact of the reuse of automotive engines. As for the examples above, they do not consider a decay in lifetime if an engine is reused but assume that a reused engine has the same lifetime as a new one.

Hence, the models considering reuse do so, by redirecting product flows without considering decays in lifetime, hoarding times, or the impact of consumer behaviour. This can pose an oversimplification which emphasizes the need for models representing circular strategies more realistically. However,

Godoy León et al. (2020) first introduced hibernating stocks in their MaTrace model shifting the perspective on material efficiency towards the user.

What can be understood as MaTrace model

Based on the large variety of present MaTrace models it appears to be legitimate to extend existing once without loosing the label. All present models have in common that they consist of three major clusters: Use, End-of-life, and production. The consideration and calculation of the closed loop in the production cluster is the aspect which clearly distinguishes MaTrace from former dynamic MFA models. Furthermore, they all trace the whereabouts of only one initial material input. In this context, dynamic stock modelling is used to consider multiple life-cycles of one inflow (one cohort) (also see Section 2.2) and not a time series input as commonly used in dynamic MFA.

I deem the way reuse is considered in MaTrace models as insufficient. Thus, the current state or the art in modeling reuse will be explained in the following section.

1.1.3. Modeling reuse

Although reuse is not the leading strategy according to the 9R framework, it is highly relevant to investigate its impact since the markets for second-hand an refurbished goods have been growing steadily in recent years (Kwarteng et al., 2018). It is a valid strategy because service lifetime and resource efficiency are closely connected. A longer lifetime increases the usage of a spent resource which increases the resource efficiency of this very item. Therefore, it is highly relevant to understand when a service life ends. Quantifying lifetimes is especially difficult for consumer goods because consumers have individual motivations to discard products. This decision can be either driven by diminished product integrity or by obsolescence (den Hollander et al., 2017). In general, a use cycle ends with the obsolescence of a product, meaning the product lost its perceived value to its owner (Box, 1983).

It is crucial to understand that there are multiple ways a product can become obsolete to its owner. The most intuitive one is functional obsolescence, meaning the product lost its functionality (Glöser-Chahoud et al., 2019). In this context it needs to considered that a loss of functionality/performance did not necessarily occur but might be perceived by the user for other reasons (Makov & Fitzpatrick, 2021). Furthermore, the actual time span within which a product stays functional can depend on the users behaviour. Batteries for example have an expected lifetime of three to five years, their actual lifetime depends on the charging habits of the user (Beaulieu, 2021).

Planned obsolescence is closely connected to functional obsolescence. Planed obsolescence is a loss of functionality by design caused for example by software updates (Glöser-Chahoud et al., 2019). Besides its functionality, it is possible that a product stops to fulfill the requirements of the owner. In this case aesthetic obsolescence describes products which are perceived outmoded, psychological obsolescence covers products which are out of fashion (Burns, 2010), and technological obsolescence describes technologies which are supplanted by superior technology (Glöser-Chahoud et al., 2019). Besides individual particularities the social environment can have an impact on the decision to discard a product. Social obsolescence describes products becoming obsolete due to laws, regulations, or stigmatization (Burns, 2010).

On this basis it can be concluded that obsolescence and service lifetimes are highly user dependent. As a consequence, it was discovered that a product may be perceived as obsolete by one user but not by another which forms the basis for reuse. Hence, a product can go through multiple use cycles which would improve the resource efficiency of this particular item.

A phenomena preventing products from entering another use cycle or waste treatment is called hibernation, meaning the storage of obsolete but functional items by the consumer (Oswald & Reller, 2011). Hibernating stocks impair resource efficiency in two ways. Firstly, items appear to be stored for a long time. Wilson et al. (2017) found that the hibernation phase of mobile phones in the UK is almost three times longer than the use phase. Hence, resources in those item are not used during this time. Secondly, due to long storage periods products are more likely to become obsolete to other potential users. This diminishes the probability for product to enter another use cycle (Glöser-Chahoud et al., 2019).

Considering the inconsistent behaviour of consumer regarding obsolescence and hoarding times it becomes obvious that service lifetimes, hibernating stocks and reuse are hard to quantify and consequently hard to model. Obtaining needed data requires conducting surveys and interviews. Modeling

reuse poses an interface from classical MFA to social science and showcases the interdisciplinarity which is required to quantify material flows in societies metabolism.

1.1.4. The reuse model

Material flows are usually modeled based on stock data, sales and estimated product lifetime. Due to the described phenomena of obsolescence, hibernating stocks and reuse create a discrepancy between high sales and low collection flows could be found.

Thiébaud et al. (2017) pointed out, that the following publications treated aspects of this issue. Williams et al. (2005) investigated the hoarding times of computers, Polák and Drápalová (2012) found hoarding times for mobile phone and Milovantseva and Saphores (2012) and Saphores et al. (2009) estimated the hoarded e-waste in households in the USA. Chung et al. (2011) estimated the storage time and amounts of among others computers, televisions, and washing machines in Hong Kong. Reuse and hibernation was also considered in multiple MFA studies (Parajuly et al., 2017; Steubing et al., 2009; Yoshida et al., 2009). However, Thiébaud et al. (2017) claim (to the best of their knowledge) that their model is the first dynamic MFA to distinguish between in-use, reuse, and hibernating stocks for electronic equipment (EE) while considering reuse and storage times.

The model itself (see Figure 1.3, in the following referred to as **reuse model**) is a cascading model considering three use and three hibernating stages where the third use and hibernating stage represent all further uses beyond (Thiébaud et al., 2017).



Figure 1.3: Reuse model system diagram by Thiébaud et al. (2017)

The original model as created by Thiébaud et al. (2017) is an inflow driven model considering multiple electronic devices. While the regional scope is clearly mentioned (Switzerland), a time frame to which the data is applicable is not provided. The model itself is a combination of multiple dynamic stocks which makes it a dynamic MFA model. Each of the stocks receives an inflow, a survival curve defines the stock decay and the outflow is distributed via transfer coefficients. Hence, the model requires extensive amount of data. Each product requires six survival curves (three for the use stocks and three for hibernating stocks) and 27 transfer coefficients to determine the fate of each outflow.

The required data is hard find thus the data was gathered in a separate publication. Thiébaud (-

Müller) et al. (2017) investigated the service lifetime, storage times, and disposal paths for ten electronic device types in Switzerland by conducting interviews and surveys. Their discussion of the results reflect the aforementioned factors contributing to uncertainty. Furthermore, it is mentioned that the data on reuse and disposal pathways might be object to subjective estimates. This limitation is in line with the results of another publication which investigated the difference in desired, expected, and actual measured lifetime. The study found that the actual measured lifetime is substantially smaller than the expected lifetime (Wieser & Tröger, 2015), this adds the dimension of consumer self perception and expectations.

Although, the model and the required data are subjected to deep uncertainty the introduced reuse model was found useful to understand alternations in consumer behaviour and use patterns (Glöser-Chahoud et al., 2019).

Since uncertainty will continue to be an important aspect of this work, the following section will clarify the uncertainties in dynamic MFA models inherent to survival curves.

1.1.5. Survival curves as example for uncertainties

Section 1.1.3 explained the underlying mechanisms causing consumer to discard product are extremely complex and non homogeneous. No matter the uncertainties, the end of life has to be represented in dynamic MFA modelling. To do so, the end of life of a whole basic population (e.g., a product group) is defined by the survival curve of a distribution. Miatto et al. (2017) showcased the difficulties and importance to select the right distribution by the example of buildings. The following explanation shall serve as an representative example for uncertainties in data and modeling choices in MFA modeling. Figure 1.4 displays the survival curves (A) and hazard rates (B) of the most commonly used distributions in MFA modelling. Survival curves represent the reliability of a product or a product group. In the context of MFA it represents the stock depletion of one single cohort (inflow into the stock at one point in time). Its value (y-axis) represents the share of material being in stock over time. The hazard rate represents the likelihood of an item failing (reaching its end of life) at a certain point in time.



Figure 1.4: Survival curves (a) and hazard rate (b) of multiple distributions. All displayed distributions have the same median value (50 % in t = 30) and the same standard deviation ($\sigma = 20$). The figure is inspired by Miatto et al. (2017).

It is important to notice that following descriptions and observations of the distributions are not all generalizable since shape and location may vary based on the parameters defining the distribution. The displayed distributions in Figure 1.4 can be categorized in right-skewed distributions (Weibull, Lod-normal, and Gamma), left-skewed distributions (Gompertz), and the normal distributions. The individual properties of the distributions need to be considered when a distribution is selected to model a stock decay. Looking at the displayed right skewed distributions it is apparent that the first value of the

survival curve is 1, meaning that the whole basic population survived till that point. This is not the case for the displayed normal distribution since its value in the survival curve is below 1. Hence, by using the normal distribution with these settings to model a survival curve it must be assume that the stock depletion starts right at the moment of material input. Regarding buildings this implies demolition of some buildings right after construction (Miatto et al., 2017) and for consumer electronics it assumes that part of the stock is discarded within a year. Also the Gompertz distribution has specific implications since it is normally used to model lifetimes of living beings. Because living beings cannot exceed a certain age (humans currently day latest around 120 years of age ("List of the verified oldest people", 2022)) the hazard function of the Gompertz distribution (see Figure 1.4 B) grows exponentially once stabilized (Miatto et al., 2017). If the Gompertz function would be applied to to buildings it had to be assumed that no building exceeds a certain maximum lifetime.

Using this knowledge, one can pick a distribution based on assumptions. Those could be the following for buildings:

- Recently build infrastructure is unlikely to get demolished in the year of construction.
- There will be a peak at in stock depletion depending on external factors as location and typology.
- Buildings surviving this peak are probably deemed to be worth preserving resulting in a lifetime far beyond average.

On this basis, it can be assumed that the log-normal distribution may be the best to model the stock decay of buildings. Miatto et al. (2017) test this assumption by fitting multiple distributions on historical data of three locations. It turned out that right skewed distributions describe stock dynamics indeed well. However, only when the underlying assumptions can be preserved throughout the lifetime. The example of the city of Salford in the United Kingdom proves the point. During the 20th century Salford went through an economic boom, followed by an economic crisis which led to the construction of inferior houses. A large share of those were demolished within a short period. As a consequence of this dynamic, the Gompertz distribution (left skewed) is most suitable in this case.

Since survival curves in dynamic MFAs are used to model an uncertain future, it may not be possible to foresee events as described. Therefore, the selection of a distribution can only be based on the knowledge available in the very moment of modelling. For those reasons assumptions on distribution shapes can turn out wrong when modelling results are compared with reality (lanchenko et al., 2020). Furthermore, data on survival curves can rarely be applied to another local or temporal context since the systems scope really matters (lanchenko et al., 2020; Yu et al., 2020).

In conclusion, it is possible to make an educated selection regarding the distribution of a survival. However, it is possible that made assumptions will turn out wrong due to unforeseen events. Furthermore, as demonstrated in Section 1.1.3 it is often difficult to obtain valid data on lifetimes. Therefore, lifetimes and their distribution pose a major source of uncertainty in dynamic MFA modelling.

1.2. Research gap and research questions

Section 1.1.3 shows that modelling reuse is complex. The data is obtained through conducting surveys and interviews, and the derived survival curves are objected to uncertainties. Presumably because of the high effort required, detailed reuse modeling as suggested by Thiébaud et al. (2017) has not been integrated into any MaTrace model yet. Therefore, the first objective of this research is to integrate Thiébaud et al. (2017) reuse model into an existing MaTrace model. This objective dictates the main research question:

How to introduce reuse into a MaTrace model and how to evaluate it?

The chosen model is the MaTrace model on cobalt in the European Union of Godoy León et al. (2020) (for more information see Section 1.3 and Section 2.1). The model created out of Thiébaud et al. (2017) reuse model and Godoy León et al. (2020) MaTrace model will be referred to as the **Reuse-MaTrace model** in the following. Once the Reuse-MaTrace model is created, the first research objective is to understand if this integration has a substantial impact. This leads to the first sub-research question:

To what extent is the outcome of the MaTrace model as described by Godoy León et al. (2020) altered, when the reuse of consumer electronics is modelled following the approach of Thiébaud et al.

(2017)?

The combination of those two models is possible since Glöser-Chahoud et al. (2019) applied Thiébaud et al. (2017) reuse model on electronics in Europe in a similar temporal scope. This endeavour requires only a minimal data gathering effort. The data present is, however, subject to uncertainty as described in Section 1.1.4. Furthermore, the lifetime of each product is represented by a survival curve and as described in Section 1.1.5 it is hard to pick the correct distribution and to find the right parameter settings. Also the data used in the MaTrace model of Godoy León et al. (2020) is not fully certain and the data quality varies depending on the parameter as they show in their previous publication (Godoy León & Dewulf, 2020). These circumstances lead to the second sub-research question:

Which input parameters introduce the most uncertainty into the Reuse-MaTrace model and what insights can be derived from analysing uncertainties?

The following section gives an introduction to cobalt as chosen material to conduct this research.

1.3. A case study on cobalt

The chosen material for this research is cobalt. Cobalt is a core material of rechargeable batteries due to its unique properties. Two current trends drive the demand for rechargeable batteries. On the one hand, the need to transition from fossil fuels to renewable energies kicked off. Electrification in the automotive industry makes electric vehicles (EV) the current, and future, main driver for demand of cobalt (Sun et al., 2019). The energy transition will require energy storage capacities (e.g., home battery storage) which will also drive the demand in the future. This leads to cobalt being considered as an energy transition metal (ETM) (Rachidi et al., 2021). On the other hand, the ongoing digitalization of industry and private life also considering trends as internet of things (IOT) and 5G technology will drive the demand for portable batteries (Desai, 2020). Certainly, this will increase the demand in the foreseeable future.

Although, the above listed driver suggest an increase in demand advancements in battery development may counteract in the future. A large electronic vehicle producer already uses cobalt-free batteries (McFarland, n.d.) while researchers use scenarios in which the cobalt demand declines in 2030 due to changes in battery technology (Tang et al., 2021). Hence, the future cobalt demand is uncertain.

The above outlined development concerns the main source of virgin cobalt. More than two thirds of the global cobalt supply is sourced in the Democratic Republic of Congo (DRC) (Farchy & Warren, 2018) where insecure mining conditions and reoccurring human rights violations such as child labor occur (Sovacool, 2019). Due to its relevance for the energy transition and the problematic origin of the virgin material I consider cobalt a material worth studying.

Looking at the publications treating cobalt using dynamic MFA (search on Scopus using the terms "cobalt AND dynamic "material flow analysis"" and "cobalt AND dynamic MFA") it is striking that the overwhelming majority has been published during the last three years. Accordingly, the publications of Pehlken et al. (2017), Bobba et al. (2019), Kamran et al. (2021), and Tang et al. (2021) entail prospective dynamic MFAs treating CRMs in EV batteries which cover cobalt as well. The regional and temporal scope vary. Furthermore, those publications also treat circular strategies as reuse, recycling and shared ownership to varying extends. In contrast Kastanaki and Giannis (2022) focuses on the future global CRM demand caused by smartphones. More broadly, Dunn et al. (2021) investigates the global CRM demand caused by batteries and considers circular scenarios.

Additionally, three publications were found conducting retrospective dynamic MFAs. Liu et al. (2021) investigates lithium and cobalt flows in China while Wang and Ge (2020) solely focus on cobalt flows in China. Godoy León et al. (2022) quantify long term cobalt stocks and flows in the European Union.

The only MaTrace model found which treats cobalt is the one of Godoy León et al. (2020) which traces the cobalt inflow of 2015 within the European Union.

On the basis of this review it can be concluded that the energy transition, especially regarding vehicles motivated a large share of the presented studies. As a consequence only few publications solely treat cobalt. Furthermore, many studies investigate the impact of circular strategies as it will be done in this research. The novelty of the Reuse-MaTrace model is the integration of an elaborate reuse model into a MaTrace model. The creation of the model and whether the integration of reuse has a

meaningful impact will be discussed in the following chapters.

2

Methodology

This chapter describes the actions taken in order to answer the research questions. The modelling approach will be described first. The MaTrace model was extended to consider the potential reuse of consumer goods (see Section 2.1). Because the model is far from trivial, the actual implementation of the model is explained (see Section 2.2). Apart from the modelling itself, the impact of the suggested changes and the model behaviour are relevant. An impact analysis to gain knowledge on this will be performed (see Section 2.3). The MaTrace model itself requires a lot of data and the model input will be inflated due to the proposed extension (also see Section 2.4.1 where the inputs are mapped out). For this reason, Monte Carlo simulations are conducted to gain knowledge about the relevance of all inputs (see Section 2.4).

2.1. Modelling approach

The foundation of the research project is a MaTrace model which considers reuse as described in Section 1.1.3. Since such a model does not exist yet, two models needed to be combined. I deemed the MaTrace model developed by Godoy León et al. (2020) to be suitable. As displayed in Figure 2.1, the model follows the generic MaTrace approach by incorporating a use, EOL and a production phase (also see Section 1.2). Exports and different losses are considered. Besides the pretreatment and recycling losses, which occur due to process efficiencies, the model considers downcycling and non-selective collection. Here downcycling refers to material which ends up in low tech products. In the context of cobalt this mainly applies to alloys containing cobalt which are then downcycled into stainless steel. In this case the material is dissipated into the technosphere (Godoy León et al., 2020) from where it is in general not recoverable (Godoy León et al., 2020). Non-selective collection refers to the misplacement of EOL products into the wrong waste bin (Godoy León & Dewulf, 2020). In contrast to the original MaTrace model, this model considers hibernating stocks (Godoy León et al., 2020).



Figure 2.1: MaTrace model system diagram by Godoy León et al. (2020)

The model is suitable due to the distribution of initial inflow as well as its consideration of cobalt, which makes it highly relevant. It considers eleven product categories: Portable batteries, mobility batteries, three types of catalysts, dissipative uses, hard metals, magnets, other metallic uses, and superalloys. 41.2 % of the whole in-flowing cobalt is allocated to portable batteries (Godoy León et al., 2020). The category portable batteries covers rechargeable batteries in consumer electronics such as laptops, cell phones, and cameras (Godoy León & Dewulf, 2020). Chapter 1 describes that the total demand for portable batteries will increase while the demand relative to the total cobalt demand may decrease. It is predicted that 30% of the cobalt used in batteries today will be used in portable batteries in 2025 (Desai, 2020).

In order to model the reuse of this portable batteries category I will extend the MaTrace model by Godoy León et al. (2020) with the reuse model of Thiébaud et al. (2017) (see Figure 2.2).



Figure 2.2: Reuse-MaTrace model system diagram

Besides the data as used by Godoy León et al. (2020) in the MaTrace model, the extended reuse part relies on data as presented by Glöser-Chahoud et al. (2019) who applied the reuse model by Thiébaud et al. (2017) on Europe. This data already shows strong tendencies towards reuse. According to Glöser-Chahoud et al. (2019) 60% of smart phones in Europe are reused, and 14 % have more than two owners. In addition to this already high reuse rate it can be assumed that the second hand market will grow in the future. Also because of the growing popularity of second hand e-commerce platforms Glöser-Chahoud et al. (2019). Figure 1.3 shows that the original reuse model also considers the fate of the disposed products. In the combined model, the outflow will be summed up and passed back to the MaTrace model which handles the fate of the disposed products.

The following section explains the functionalities of the combined model.

2.2. Model implementation

Figure 2.2 displays, the start of model, the use phase has two inputs; the initial products and products produced from secondary materials in the production phase. Thus, there is only one inflow from outside the system boundaries. In order to determine the inputs to the use phase of the second year, the whole model has to be executed for the initial year first. The material leaving the use phase has to pass through the end-of-life phase. Retrieved secondary material will be used in the production phase to produce new products. A share of those products become the inflow into the use phase of the second year. Therefore, the model has to be run iteratively.

In the first iteration, the initial inputs enter the system and get split up. Portable batteries enter the cascaded reuse model, all other categories are treated as in the MaTrace model: they enter a use stage and one hibernating stage afterwards. Within the reuse model the inflow is split into seven products with individual data on service lifetime, hibernating time (represented by survival curves), and reuse (represented by transfer coefficients) (Also see Chapter 3).

All stocks, in-use and hibernating, are represented by one cohort matrix per product (in reuse model) or per product category (as in MaTrace model by Godoy León et al. (2020)). A cohort matrix represents the net-inflow, the stock itself, and the depletion of a stock. It is created by the inflow and the survival curve.

	0	1	2	3	4	5	6	7
0	0	0	0	0	0	0	0	0
1	0	0.005779	0	0	0	0	0	0
2	0	0.004155	0.009849	0	0	0	0	0
3	0	0.002126	0.007081	0	0	0	0	0
4	0	0.000853	0.003623	0	0	0	0	0
5	0	0.000279	0.001454	0	0	0	0	0
6	0	7.59E-05	0.000475	0	0	0	0	0
7	0	1.75E-05	0.000129	0	0	0	0	0

Table 2.1: Cohort matrix representing the 2nd in-use stock of smart phones until the third iteration

Table 2.1 displays the cohort matrix of the second in-use stock (first reuse stock) of smartphones after the third iteration. The diagonal of the matrix holds the stock addition in each year, which is not necessarily equal to the inflow, since inflow may already partly decay during the first year. This depends on the distribution used for the survival curve (also see Section 1.1.5). The inflow multiplied with the survival curve defines one column. Hence, one column provides the knowledge, how much was added to a stock in one year (entry on the diagonal) and how much of this addition is in stock in the following years. Due to this circumstance, one finds the total material in stock of a year by summing a row (derived from B. Müller (2006)). Because of the iterative nature of the model, the matrix is not fully filled after two iterations. Furthermore, since the stock outflow is put into the MaTrace model, it might be that a former outflow flows back into the stock at some point. Table 2.2 shows the cohort matrix of the second in-use stock of smart phones when the model is run for eight iterations.

	0	1	2	3	4	5	6	7
0	0	0	0	0	0	0	0	0
1	0	0.005779	0	0	0	0	0	0
2	0	0.004155	0.009849	0	0	0	0	0
3	0	0.002126	0.007081	0.009332	0	0	0	0
4	0	0.000853	0.003623	0.00671	0.007348	0	0	0
5	0	0.000279	0.001454	0.003433	0.005283	0.005278	0	0
6	0	7.59E-05	0.000475	0.001378	0.002703	0.003795	0.003532	0
7	0	1.75E-05	0.000129	0.00045	0.001085	0.001942	0.00254	0.002184

Table 2.2: Cohort matrix representing the 2nd in-use stock of smart phones until the eighth iteration

The first column of the matrix only holds zeros. This suggests that there is no inflow to the second use-stock in the first year. Only in the second iteration the first in-use stock starts to deplete and part of the phones leaving the first in-use stock enter the second.

Within the reuse model in the use phase transfer coefficients define which share of a stock outflow goes to either the next use-stock, the hibernating stock, or to disposal. Since the MaTrace model incorporates the waste treatment process including losses and exports, the reuse model considers only one disposal way resulting in the EOL products. In order to suit the MaTrace model, the individual product outflows of the reuse model are summed up into the category *portable batteries* again.

The clusters End-of-Life and Production are equivalent to the MaTrace model by Godoy León et al. (2020). Both clusters are defined by transfer coefficients and allocation matrices. Furthermore, a infinite geometrical series emerges due to the close loop in the production cluster. This series is solved analytically (see Section 1.1.2).

2.3. Impact analysis

The impact analysis serves the purpose to investigate how the output of the MaTrace model is altered due to the modification. I chose the total in-use stock as the measure of comparison since this model output is closely related to resource efficiency (see Section 1.1.2). The impact analysis consist of the following two steps: Comparison to the MaTrace model by Godoy León et al. (2020) and the impact of reuse.

2.3.1. Comparison to the MaTrace model created by Godoy León et al. (2020)

Firstly, it will be investigated how the in-use stock changes once the reuse model is added. In a second step the impact of the usage of an altered data set (see Section 3) will be examined. Based on this analysis it will be possible to answer, whether the consideration or reuse is of relevance at all regarding this specific case study.

2.3.2. Impact of reuse

Secondly, the impact of the consideration of reuse is evaluated in greater detail. To do so, the part of the model considering reuse (see marked area Figure 2.3 A) is altered. In order to create a baseline, the reuse model is reduced to showcase single use only (see Figure 2.3 B). Then the model is run with two use stages (see Figure 2.3 C) and then with three (see Figure 2.3 D).



Figure 2.3: Scenarios for the reuse impact analysis; (b) considering single use, (c) considering one reuse phase, and (d) considering the whole reuse model. The red dotted line in (a) marks the location of the use phase in the whole system.

The results of this analysis will allow to draw conclusions on the impact of the consideration of reuse on the in-use stock.

2.4. Monte Carlo simulations

Monte Carlo methods are used in different scientific fields for various purposes (Dimov, 2008) and are formally defined as: "methods of approximation of the solution to problems of computational mathematics, by using random processes for each such problem, with the parameters of the process equal to the solution of the problem." (Dimov, 2008, p.1)

In summary, the inputs of a model which have an effect on the solution (or model output) are varied in a random fashion. When applied on MFA models, selected or all inputs are defined by a probability distribution. Random inputs are generated on the basis of those distributions which are then fed into the model. This process is repeated multiple times. As a result, one obtains probability distributions of the output data (Sonnemann et al., 2003). It is important to understand, that inputs are not varied arbitrarily randomly but that the inputs are generated on the basis of the assigned probability distribution). This also entails to set limits of the number generator to a plausible range. Furthermore, not all values have the same probability to be generated.

Besides the more obvious result of the probability distributions of outputs, one can derive further conclusions by analyzing the results. In the context of MFA, Monte Carlo simulations are used to analyze the uncertainty and sensitivity of models (Fishman et al., 2018). In this case that is also the purpose. The presented combined model is unique and unproven. Although, the data was mainly collected from previous peer reviewed publications, the authors of these publications recognize the weaknesses of the data quality themselves. Godoy León and Dewulf (2020) evaluated the data quality of each input used in the MaTrace model by Godoy León et al. (2020) while Thiébaud et al. (2017) stressed that the data for the reuse model has to be obtained by interviews and surveys. Hence, knowing which inputs have the largest impact on relevant outputs is important information in case practitioners consider to use the model in the future. This way, users can know which inputs require a high certainty (i.e. data quality) and which inputs require less attention.

Even though, the method appears to be established in MFA modelling, approaches on how to assign probability distributions to inputs vary among publications. It seems common to assign distributions on

the basis of the known uncertainty of the inputs (Cai et al., 2022; Cao et al., 2018; Fishman et al., 2018; Glöser et al., 2013; Gottschalk et al., 2010; Montangero & Belevi, 2008; Thiébaud et al., 2018). But the derived distributions vary immensely. Fishman et al. (2018) used exclusively normal distributions, while Thiébaud et al. (2018) use triangular and normal distributions for inputs with higher certainty. Montangero and Belevi (2008) derives most input distributions including parameters from existing literature, and Gottschalk et al. (2010) uses log-normal, triangular or uniform distributions depending on input characteristics.

On the one hand this poses the challenge to execute Monte Carlo simulations without clear guidelines. On the other hand, it allows to develop a framework according to the needs of the model and researcher. In order to develop this framework, the following section maps out all inputs of the model. On the basis of this knowledge a framework to assign distributions will be developed.

2.4.1. Model input mapping

The data inputs to the model need to be mapped to create an overview. Furthermore, the total number of inputs is calculated in this section.



Figure 2.4: Reuse-MaTrace model and required data (adopted from Godoy León et al. (2020))

Figure 2.4 displays the system and data required to run the model, which are represented by boxes holding mathematical symbols. Each symbol has one or more indices (small letters) indicating whether this data is required for each product (index *i*, consumer electronics as e.g., laptops), product group (index *k*, e.g., hard metals), use cycle (index *l*, referring to use- and reuse- cycles not recycling), recycling process (index *r*, recycling processes as present in MaTrace model), or secondary materials (index *m*, material created by recycling processes). If multiple indices are given, the data is required for every product in every use-cycle). I calculated the number of individual inputs on the basis of Figure 2.4. Therefore, a norm must be defined which gives the number of elements in a vector or matrix of the dimensions $\mathbb{R}^{p\times q}$:

$$|A||_{R} = \sum_{j=1}^{p} \sum_{h=1}^{q} 1$$
(2.1)

Considering the seven products (index i), 11 product groups (index k), three use cycles (index l), two recycling processes (index r), and two secondary materials (index m) one finds the following table

by applying the $||A||_R$ norm:

Vector or Matrix	Dimension	Number of elements
ε	$k \times 1$	11
δ	$i \times 1$	7
α	$i \times l - 1$	14
β	i imes l	21
γ	$i \times l - 1$	14
ψ_{EOL}	$k \times 1$	11
σ	$k \times 1$	11
λ_{PT}	$k \times 1$	11
В	$k \times r$	22
λ_R	$r \times 1$	2
ψ_R	$m \times 1$	2
D	$k \times m$	22
λ_P	$k \times 1$	11
λ_M	$k \times 1$	11
ξ_P	$k \times 1$	11
ξ_M	$k \times 1$	11
ω_P	$k \times 1$	11
ω_M	$k \times 1$	11
ψ_P	$k \times 1$	11

Table 2.3: Dimensions and number of input parameters

The sum of all elements is 225, while the part of the model which treats the reuse of consumer electronics requires 56 inputs. However, Table 2.3 does not consider the number of the required survival curves which is displayed in the following table:

Survival curve	Number of survival curves
S_U	$i \cdot l + k - 1 = 31$
S_H	$i \cdot l + k - 1 = 31$

Table 2.4: Number of required survival curves

In the combined model, all survival curves are defined by Weibull distributions, described by two parameters. Thus, the original model has $62 \cdot 2 + 225 = 347$ input parameters. If the distribution describing the survival curves is changed during the course of the experiments, this number can vary.

Hence, the part of the model which treats the reuse of consumer electronics requires 56 inputs and 42 survival curves (82 additional inputs).

2.4.2. Framework

The number of input parameters is too large to define a specific distribution for each, which means that an alternative, feasible approach needs to be applied. This approach consists of two steps. Firstly, each input is going to receive a score mainly based on uncertainty and where known by sensitivity. Secondly, these scores will be used to assign a distribution to each input.

Input scoring

Scoring is a commonly used practise in the field of IE to compare things as the impacts of multiple environmental interventions. Often multiple categories are scored on the basis of qualitative or quantitative information and one main score is aggregated (Duc et al., 2022; van Berkel et al., 1997). The main drawback is the low repeatability of the score assignment (Powell et al., 1997). Therefore, I explain how scores were assigned in the following.

The model contains five different input data types: the initial inflow, transfer coefficients, efficiencies, the Weibull scale, and the Weibull shape. The first four types can receive an uncertainty score from 0 to 5. Here, 0 represents absolute certainty and 5 stands for very high uncertainty. An example for

an absolute certain input is the recycling efficiency of dissipative uses. In MaTrace models dissipative uses are the use of material in e.g., medicine or pesticides or regarding cobalt, the use of the material as pigment. So, the material cannot be recovered by definition. Therefore, the recycling efficiency for cobalt recovery from dissipative uses is 0 with absolute certainty. The score of 5 was assigned for inputs which are based on my own assumptions.

The Weibull shape parameter β can only receive scores from 0 to 2. The reason for this is its known sensitivity. The shape of the Weibull distribution can change severely when the parameter varies. Figure 2.5 A displays the survival curves of smartphones in the first service phase considering multiple shape parameters. When $\beta > 1$ as for the original shape ($\beta = 1.7$) the failure rate increases over time (see Figure 2.5). As a result, few smartphones fail during the fist two years and most of them drop out of service between the second and the third year. This behaviour changes for $\beta \approx 1$ which leads to a more constant failure rate. For $\beta < 1$ the failure rate is high at the beginning and decreases over time. As it can be seen in Figure 2.5, this leads to an early failure behaviour. Less than half of the smart phones survive the first two years (KIzIlersü et al., 2018). For this known sensitivity of the Weibull shape, the uncertainty scores are limited to the mentioned range.



Figure 2.5: Survival curve (a) and probability density function/failure rate (b) of Weibull distribution for multiple shape parameter β

The inputs which come from the MaTrace model by Godoy León et al. (2020) are scored on the basis of a previous publication of the same authors. Godoy León and Dewulf (2020) developed a framework to asses the quality of CRM related data. This framework was applied to the MaTrace model which is used in this thesis. Hence, the data quality of each input to the MaTrace model was scored from "Very high data quality" over "High data quality", "Low data quality", and "Very low data quality" to "No data reported" (Godoy León & Dewulf, 2020, Table 3). This data quality stands in direct relation to data uncertainty. High quality data is more reliable and therefore more certain. The data quality evaluation of Godoy León and Dewulf (2020) is translated as follows:

- · Uncertainty score 1: "Very high data quality"
- Uncertainty score 2: "High data quality"
- Uncertainty score 3: "Low data quality"
- · Uncertainty score 4: "No data reported"

The category "Very low data quality" does not refer to an uncertainty score, since this category was not assigned to any input. Furthermore, the category "No data reported" was matched with the uncertainty score 4, since Godoy León et al. (2020) filled in missing data based on their research,

assumptions, and expertise. Because I assume that the validity of their data assumptions is more valid than the data which bases on my own assumption, this kind of data did not receive the lowest score.

The uncertainty of the data which originates from Glöser-Chahoud et al. (2019) application of Thiébaud et al. (2017) reuse model is scored by my own judgement. Regarding the first use and storage phase it is assumed that the survival curves defined by Glöser-Chahoud et al. (2019) have a high certainty as they are based on multiple empirical studies. Accordingly, the shapes receive a uncertainty score of 1 and the scales receive an uncertainty score of 2. Since the survival curves of the products e-bikes, power tools, and others base on own research, they are scored lower. Weibull scales received a score of 4 and Weibull shapes a score of 2. The transfer coefficients as suggested by Glöser-Chahoud et al. (2019) are derived from previous research (survey and interview studies) thus they are assumed to be less certain and receive an uncertainty score of 3. The transfer coefficients for e-bikes, power-tools, and other products are scored with 4 since they are based on literature and derived assumptions (see Chapter 3).

For the second re-use and storage stage I assumed that the data is less certain than the data of the first stage. The data of the reuse model largely depends on surveys and interviews (Thiébaud et al., 2017). I assumed that the information gathered becomes vague in the second stage since part of the information provided is less certain. It is for example less likely that owners of second hand articles know the exact purchase date because they may not have a receipt. Furthermore, they might not be aware of all previous owners and may mistake a third hand product for a second hand product. In order to account for those kind of added uncertainty, I lowered all all uncertainty scores by one.

Regarding the third re-use and storage stage I assumed that the data is not certain at all. The aforementioned uncertainties are present as well. Furthermore, owning a third hand product is less likely than owning a second hand product. Therefore, the pool of possible survey and interview participants must have been smaller. All inputs on this stage are thus rated with the highest uncertainty possible, which is 2 for Weibull shapes and 5 for all other parameters.

Appendix B shows assigned uncertainty scores of all inputs.

Distributions

As explained above, the uncertainty scores are used to assign distributions to the inputs. How this is done will be explained in the following. The fundamental idea is to assign broader distributions to uncertain inputs and narrow distributions to inputs with a high certainty. This results in the following pairing for this Monte Carlo simulation:

- · Uncertainty score 0: The input value is not changed at all
- · Uncertainty score 1: The input value is altered by a normal distribution
- · Uncertainty score 2: The input value is altered by a broader normal distribution
- · Uncertainty score 3: The input value is altered by a triangular distribution
- · Uncertainty score 4: The input value is altered by a broader triangular distribution
- Uncertainty score 5: The input is very uncertain and therefore altered by a uniform distribution (Gottschalk et al., 2010)

The specific parameters of the distribution depend on the type of input data. The initial inflow, transfer coefficients, and efficiencies can take values from 0 to 1 while the Weibull scale and shape can take theoretically arbitrary values larger than 0. Therefore, those groups of input are treated differently.

Regarding the Weibull scale and shape, the distribution does not alter the input value directly but the distributions generate a multiplier for the input. The multiplier can take values from 0.5 to 1.5. So, the input value can be 50% larger or smaller than the initial value. The distributions for uncertainty scores 3 to 5 are limited to this range. If a normal distribution for the uncertainty scores 1 and 2 generates a value out of this range, a new value will be generated until it is within this range. Figure 2.6 A displays the distributions for this case:

- Distribution for uncertainty score 1: Normal distribution with $\mu = 1$ and $\sigma = 0.05$
- Distribution for uncertainty score 2: Normal distribution with $\mu = 1$ and $\sigma = 0.1$
- Distribution for uncertainty score 3: Triangular distribution limited from 0.6 to 1.4 with the peak at 1
- Distribution for uncertainty score 4: Triangular distribution limited from 0.5 to 1.5 with the peak at 1
- Distribution for uncertainty score 5: Uniform distribution limited from 0.5 to 1.5



Figure 2.6: Distributions of survival curve multiplier for Monte Carlo simulations. (a) shows the distribution without limitations. (b) shows the distributions when the smallest possible multiplier is limited to 0.7.

As explained above, the Weibull scale β changes the shape of the survival curve severely when it changes from $\beta > 1$ to $\beta < 1$. Therefore, it was decided that Weibull scales which are initially larger than 1 cannot take values smaller than 1. To ensure this, the smallest possible multiplier for which $\beta \leq 1$ is calculated and used as the lower limit for all distributions (see Figure 2.6 B, here the smallest possible multiplier is 0.7). The normal distributions generate values below this threshold, new values are generated until all values are within the allowed range.

The same applies for the Weibull shapes: A shape below 1 would suggest that most of the stock depletes during the first year which is, looking at the data, for most products and cases an unreasonable assumption. Furthermore, the Weibull survival curve always has a value of 1 for the input year. Hence it cannot project stock depletion in the first year. Furthermore, transfer coefficients define in most cases the inflow to the stocks. Therefore, there is no need for the survival curve to represent inflows that immediately exit.

In contrast to the Weibull parameters the distributions for initial inflow, transfer coefficients, and efficiencies take the actual input value as a mean for the normal distributions and as peak for the triangular distributions. The general settings are as follows:

- Distribution for uncertainty score 1: Skewed or folded normal distribution with $\mu = 1$ and $\sigma = 0.05$
- Distribution for uncertainty score 2: Skewed or folded normal distribution with $\mu = 1$ and $\sigma = 0.1$
- Distribution for uncertainty score 3: Triangular distribution limited from 0.6 to 1.4 with the peak at 1
- Distribution for uncertainty score 4: Triangular distribution limited from 0.5 to 1.5 with the peak at 1
- Distribution for uncertainty score 5: Uniform distribution limited from 0.5 to 1.5

Figure 2.7 A displays the distributions when the input value is 0.5, the distributions for other values change. Figure 2.7 B shows the distributions for an input value of 0.2. It shows that the distribution for uncertainty score 1 changes to a right skewed normal distribution. The reasoning behind this implementation is that it might be interesting for small values to test how the model behaves for larger values rather even smaller inputs.



Figure 2.7: Distribution of initial inflow, transfer coefficients, and efficiencies for (a) an input value of 0.5 and (b) an input value of 0.2

The skew factor *a* of these two distributions was determined empirically. The target was, to keep the value of the probability density function of the skewed normal distribution (S for x = 0 smaller or equal to the value or the original normal distribution for $\mu = 0.5$ at x = 0, which is a value very close to 0¹:

$$S(x=0|\mu,\sigma^2=0.05^2,a) \le \mathcal{N}(x=0|\mu=0,\sigma^2=0.05^2)$$
 (2.2)

In order to avoid a numerical approximation of *a* for each input, it was approximated by a linear function once the boundary values were determined. The boundary for the location of the skewed normal distribution (see SciPy, 2022) for the uncertainty score 1 are 0.05 and 0.95, and 0.1 and 0.9 for the uncertainty score 2. It was found that the skew factor *a* becomes extremely large beyond those boundaries. Therefore, the skewed normal distribution is replaced by a folded normal distribution which avoids this unwanted behavior. This can be seen in Figure 2.7 B for the uncertainty score 2 (green line). Although, the peak of the folded distribution changes, it needs to be disclaimed that the mean of the distribution does not necessarily match the actual input.

The behavior of the triangular distributions is less complicated. The distribution for the uncertainty score 4 moves symmetrically along the x-axis until its corner reaches either 0 or 1 (see Figure 2.8).

¹Equation 2.2 holds true for the distribution for the uncertainty score 1, for uncertainty score 2 σ has to be changed to $\sigma^2 = 0.1^2$



Figure 2.8: Distribution of initial inflow, transfer coefficients, and efficiencies for an input value of 0.4

Once this happens, the peak of the triangular distribution is set to the location of the original input value. This is also the general behaviour of the triangular distribution assigned to inputs with the uncertainty score 4 (see Figure 2.7 B and 2.8)

Examplary histograms of the resulting input distributions are shown in Appendix C. As those examples show, not all histograms follow the defined distribution. This is due to the fact that transfer coefficients which define the forking of one flow have to sum up to 1. This way, mass conversation is insured. This is done after the random numbers are generated. It is necessary to generate a numbers for all transfer coefficients to see the contribution to uncertainty of each single input.

2.4.3. Evaluation of Monte Carlo Simulations

The results of the Monte Carlo simulations which are going to consist of the results of n = 10,000individual runs are going to be used to attribute the uncertainty in the model to the inputs and to perform a sensitivity analysis following the example of Fishman et al. (2018). The first step in the evaluation is to find the correlation between all inputs and one or several selected outputs (e.g. the total use stock). Although, the Pearson correlation obtains more precise results, it cannot be used because not all of the inputs are normally distributed (see above) which is a necessary condition. Therefore, the Spearman rank correlation is used (Gauthier, 2001). Therefore, the inputs have to be ranked as presented in the following table:

	Input		Result					
-			Year 0			 Year 35		
	Value	Rank	Value	Rank	d	 Value	Rank	d
run 1	0.89	2	0.65	3	1	0.69	1	-1
run 2	0.85	3	0.66	2	-1	0.68	2	-1
:	÷	÷	÷	÷	÷	:	:	÷
run 1000	0.91	1	0.72	1	0	0.63	3	2

Table 2.5: Example Monte Carlo inputs and results including value and rank

The highest value receives the rank 1, the next largest value receives the rank 2, and so on. In the following equations, inputs are represented by a and results by b, and the index i represents inputs, the index r represents runs, and the index j represents years. To obtain the correlation, the difference between the ranks of the input and the results have to be calculated (Cao et al., 2018):

$$d_{irj} = b_{irj} - a_{ir} \tag{2.3}$$

It is important to recognize, that the results change with each year while the input of the model stays constant. Using d, the Spearman's rank-correlation coefficient can be calculated where n is the number of runs (Cao et al., 2018):

$$\rho_i j = \frac{1 - 6\sum_{r=1}^n d_{irj}^2}{n^3 - n} \tag{2.4}$$

On this basis, the uncertainty contribution of each input regarding the selected output can be calculated. This is expressed by the normalized squares of the Spearman's rank-correlation coefficients (Cao et al., 2018):

$$C_{i} = \frac{\rho_{j}^{2}}{\sum_{i=1}^{n} \rho_{ij}^{2}}$$
(2.5)

So, C_i represents the relative contribution of an input to the variance of the considered result (uncertainty) (Cao et al., 2018).

3 Data

This chapter describes the origin of the used data. Assumptions were made since existing data sets were extended and data on reuse is scarce. In the attempt to explicitly point out all assumptions, they were highlighted in italic.

3.1. Data as present in the models

The used MaTrace model of Godoy León et al. (2020) takes the year 2015 as initial input year. Considering the eventful history of recent years and the mentioned drivers influenced by climate mitigation policies, it is fair to assume that the situation has changed drastically (in the year 2022). However, the choice of the initial input year is based on the previous work of the researchers.

Godoy León and Dewulf (2020) developed a framework to assess the quality of collected data and applied the framework on cobalt. Therefore, they consulted over 330 sources and withdraw data from 76. Each obtained value received a Data Quality Rating (DQR) which is based on four to five criteria. Based on this research, the most reliable data was selected to conduct the MaTrace model (Godoy León et al., 2020).

The data used for the reuse model is withdrawn from the work of Glöser-Chahoud et al. (2019) who applied the model of Thiébaud et al. (2017) on Europe. Since they use the year 2015 as the initial year as well, the data of both models is compatible regarding temporal and regional scope. The data set which combines the data from Godoy León et al. (2020) and Glöser-Chahoud et al. (2019) will be referred to as **Combined Data Set** in the following.

3.2. Extending reuse data

The reuse model of Glöser-Chahoud et al. (2019) tries to cover consumer electronics in Europe by considering smartphone, mobile phones, tablets and laptops. Although these devices cover the largest share, they do not cover the whole landscape of devices requiring portable batteries. Furthermore, it poses a rather homogeneous group regarding their lifetime distributions. In order to add diversity, I decided to extend the data by three additional categories: cordless power tools, E-bikes and others. This extended data set will be referred to as **Extended Data Set** in the following. The distribution of the initial inflow will base on Pillot (2012) who provides the market share of different devices in *MWh* battery capacity. *It is assumed that the proportional capacity reflects the relative cobalt content.*

Cordless power tools are used by professionals and consumers this includes tools such as drills and saws, as well as gardening tools such as mowers and hedge trimmers (Union, 2006). Cordless household tools will be considered in the same category.

The category Others will entail less significant electronics such as MP3 players, cameras and toys.

3.3. Data on E-bikes

Currently, data on E-bikes is scarce, contemporary studies focus on customers who recently bought an E-bike and not on long term ownership patterns (Fyhri & Beate Sundfør, 2020). Additionally it needs to

be considered, that data regarding the use and reuse of E-bike batteries needs to be derived and not necessarily data on E-bikes themselves.

The only hard data available is the average ownership time of a bicycle, 8.3 years (Balton, 2022) and the average lifetime of an E-bike battery which is 4.12 years (with an Weibull shape of 4.34) (Guo et al., 2021). This means, that an E-bike owner will most likely use multiple batteries throughout the lifetime of a bike. *I assumed that this circumstance leads to a decreased probability for E-Bike batteries to be reused. Therefore it is assumed that a total of 10% of firstly used batteries will be reused.*

Regarding the share of devices entering the hibernating stock, it is considered, that size plays an important role. Small items are more likely to be stored than large items (Thiébaud (-Müller) et al., 2018). Since there is data on TV hibernation (Thiébaud et al., 2017) and since a TV is the device which comes closest in size of all available ones to an E-bike battery, I assumed that E-bike batteries are hoarded to the same extent as TVs.

Those assumptions inspired the transfer coefficients as presented in Table 3.1.

From	То	Coefficient
	Secondary use	0.07
First service life	First storage	0.22
	Disposal	0.71
	Tertiary use	0.03
Second service life	Second storage	0.26
	Disposal	0.71
Third convice life	Third storage	0.29
THILD SELVICE IIIE	Disposal	0.71
Eirot atorogo	Secondary use	0.14
First storage	Disposal	0.86
Second storage	Tertiary use	0.7
Second storage	Disposal	0.93

Table 3.1:	E-Bike	transfer	coefficients
------------	--------	----------	--------------

The estimated lifetime of E-bike batteries are known and the assumption is established that only few batteries are going to be reused. *Therefore, the given average lifetime is assumed for the first use. The Weibull scale is diminished to account for batteries leaving the stock and which are reused.*

The lifetime for the second use is the same since it is assumed that customers buying E-bike batteries second hand have a higher tolerance towards a diminished capacity. The third average service lifetime is set to 2 years. This is based on the underlying assumption that batteries may not be fit for purpose very long once they enter their third service life.

The hibernating times are made up based on the assumption that storage times of perishable products should not be longer than their service time. The Weibull shapes are assumed to be the same ones as for TVs.

	Mean lifetime	Weibull shape
First service life	4.12	3
Second service life	4.12	2.5
Third service life	2	1.5
First storage	2	1.7
Second storage	1.5	1.59
Third storage	1.5	1.59

Table 3.2: E-Bike lifespans

3.4. Data on cordless power tools

The estimated lifetime of batteries used in cordless power tools was found to be 10 year (Mudgal et al., 2011). Therefore, i assumed that in private ownership the lifetime of the tool exceeds the lifetime of the battery. Based on this I assumed that most batteries go to disposal, few to storage and only little to reuse.
From	То	Coefficient
	Secondary use	0.05
First service life	First storage	0.15
	Disposal	0.8
	Tertiary use	0.02
Second service life	Second storage	0.2
	Disposal	0.78
Third convice life	Third storage	0.3
	Disposal	0.7
First storage	Secondary use	0.1
First storage	Disposal	0.9
Second storage	Tertiary use	0.05
Second storage	Disposal	0.95

The following transfer coefficient table is based on those assumptions.

 Table 3.3:
 Cordless power tools transfer coefficients

Regarding the Weibull shape of the lifetime it is set to be 1.5 for the first service life while it is smaller for the followings based on assumptions hereafter. There are high frequency equipment users and others who use it only rarely. Hence there is a discrepancy in lifetime among users. To model this, the Weibull scale was reduced considering earlier failures. The storage periods are assumed to be longer than for E-bikes, since it is assumed that tool owners have more space to store EOL products.

	Mean lifetime	Weibull shape
First service life	9	1.5
Second service life	5	1.2
Third service life	2	1.1
First storage	3	1.7
Second storage	2	1.59
Third storage	2	1.59

Table 3.4: Cordless power tools lifespans

3.5. Data on other cordless electronics

The category Others includes e.g., cameras, mp3 players, and similar devices. *I assumed that those devices are rather small and easy to store when they reached their EOL. Following this, I assumed that they follow the use and storage behaviour of mobile phones as described by Thiébaud et al. (2017).* Transfer coefficients

From	То	Coefficient
	Secondary use	0.05
First service life	First storage	0.15
	Disposal	0.8
	Tertiary use	0.02
Second service life	Second storage	0.2
	Disposal	0.78
Third convice life	Third storage	0.3
	Disposal	0.7
First storage	Secondary use	0.1
T IISt Storage	Disposal	0.9
Second storage	Tertiary use	0.05
Second Storage	Disposal	0.95

Table 3.5: Others electronics transfer coefficients

The same assumption is applied to the lifetimes.

	Weibull scale	Weibull shape
First service life	3.27	2.13
Second service life	3.59	2.17
Third service life	3.59	2.17
First storage	2.16	1.17
Second storage	3.9	1.46
Third storage	3.9	1.46

Table 3.6: Others electronics lifespans

4 Results

In this chapter the results of the research will be presented following the order outlined in the methodology chapter (see Chapter 2 Methodology). First, the general model output will be described (see Section 4.1). As the model is largely based on the work of other authors, this is followed by the results of the impact analysis. I will present how the model and data extension alter the most important measure the total in-use stock (see Section 4.2.1). After this, the altered model (the Reuse-MaTrace mode) and data set (the Extended Data Set) will be used for all further experiments. Also for the second part of the impact analysis displaying the impact of adding a second and a third use cycle to the system (see Section 4.2.2). The last part of this chapter covers the results of the Monte Carlo Simulations which give insight into the uncertainties of the model inputs (see Section 4.3).

4.1. General model output

In this section, the general model output is described which refers to the model as it was presented in Section 2.1 and Section 2.2 using Extended Data Set as it was described in Chapter 3. The general model output is best described by the following figure because it represents the stocks and accumulated outflows. This means that it displays the fate of the in-flowing cobalt in 2015 over the course of the next 35 years.



Figure 4.1: Distribution of stocks and and accumulated outflows based on the cobalt inflow in 2015. The black dashed line marks the total in-use stock. Areas in red with white stripes mark losses. First, second and third use of portable batteries is represented by different shades of blue.

The figure is an area chart in which stocks and accumulated outflows are stacked on top of each other. The top layer always reaches 100 % which indicates that the graph covers all outflows considered in the model. The red dashed line illustrates the percentage of the material which can be found in the in-use stock. All areas below this line represent the in-use stock of the different product categories. The product category *Portable batteries* is broken down into first, second, and third use since this is the product category on which the reuse model was applied (see Figure 4.2 for breakdown into products).

The black areas with white stripes represent losses. Downcycling refers to the conversion of metals and alloys to stainless steel (also see Section 2.1 and Godoy León et al. (2020)) and is considered to be a loss. The two areas in between are hibernating stocks and exports. Those are not part of the in-use stock nor the losses since exported materials and goods have the potential to be reused within another geographical scope. The same holds true for hibernating stocks.

Figure 4.1 allows to draw conclusions regarding the overall dynamic of the model in combination with the used data. Looking at the losses, it is apparent that non-selective collection makes up by far the largest share of the losses (about 43 % of the former inflow in 2050) while recycling losses (pre-treatment and recycling itself) account for only about 13 % of the former inflow in 2050. Production losses appear to be very small in comparison.

Looking at exports and hibernating stocks it can be seen that hibernating stocks build up in the beginning and slowly vanish towards the year 2050. The area of exports is constantly increasing until it becomes, with about 30 %, the second largest in 2050.

Regarding the stocks of in-use product categories, it can be seen that most stocks deplete constantly. *Dissipative uses* are the only exception from that behaviour, they make up 9 % of the inflowing material in the year 2015 and the stock increases until it peaks with 9.8 % in the year 2021. It depletes from there onwards and it can be observed that the stock of portable batteries depletes comparatively quick until it is almost vanished in the year 2030. Although, the secondary in-use stock of *portable batteries* covers a clearly visible area it is important to recognize that the stock of portable batteries experiencing a third use-cycle can barely be seen.

Figure 4.2 was created in order to investigate the dynamic of the *portable batteries* stock. It is an area chart displaying the in-use stock of the *portable batteries* split up in products and use cycles.



Figure 4.2: Distribution of the in-use stock of portable batteries among products of the Reuse-MaTrace model using the Extended Data Set which consists of the data of Godoy León et al. (2020), Glöser-Chahoud et al. (2019), and other collected data (see Chapter 3). The individual use cycles are separated by texture and saturation.

Besides the general stock distribution over products, several observations can be made. Firstly, it can be seen that the second use phase holds significant shares for some product groups such as smartphones and tablets. For other product groups such as power tool and e-bike batteries, the share of the second use-phase is barely visible which is in line with the assumptions made in Chapter 3. Furthermore, the share of the products in their third use phase can only be seen when zoomed in. This observation gave the impulse to closer investigate the impact of the consideration of multiple use cycles (for results see Section 4.2.2).

4.2. Impact analysis

In the following, I will investigate what impact it has to incorporate the reuse model into the existing MaTrace model as created by (Godoy León et al., 2020). I explained in Section 2.3, that I will investigate how the total in-use stock changes. First I will focus at the changes when the reuse model is integrated into the MaTrace model (considering Combined Data Set and Extended Data Set), and secondly, on how much the second and third use cycle add to the total in-use stock.

In order to put the deviation into perspective I decided to compare the points in time when the total in-use stock hold 75%, 50%, and 25% of the inflow in 2015.

4.2.1. Model extension

As outlined above, the first impact to investigate is the extension of the MaTrace model by Godoy León et al. (2020) by the reuse model. Figure 4.3 displays (a) the total in-use stock and (b) the in-use stock of *portable batteries* of the MaTrace model by Godoy León et al. (2020) and the Reuse-MaTrace model using Combined Data Set and Extended Data Set. The vertical lines mark the instances in time when the in use-stock holds 75 %, 50 %, and 25 % of the in-use stock.



Figure 4.3: Comparison of the total in-use stock (a) and the portable batteries in-use stock (b) of the MaTrace model by Godoy León et al. (2020), the Reuse-MaTrace model using the Combined Data Set (green line, consisting of data of Godoy León et al. (2020) and Glöser-Chahoud et al. (2019)), and the Reuse-MaTrace model using the Extended Data Set (red line, the data set includes additional products for the product category portable batteries, see Chapter 3). The vertical lines marks the time when the representative stock holds only 75 %, 50 %, and 25 % of the initial inflow.

Table 4.1 holds the stock depletion instances regarding the total in-use stock (as in Figure 4.3 A).

Instant when the	7	5 %	5	0%	2	5 %
in-use stock is depleted to:	Year	Deviation	Year	Deviation	Year	Deviation
MaTrace model by	2018.0	0.0	2020.0	0.0	2020 1	0.0
Godoy León et al. (2020)	2010.0	0.0	2020.9	0.0	2029.4	0.0
Reuse-MaTrace model	2018.0	0.0	2022.2	13	20294	0.0
(Combined Data Set)	2010.0	0.0	2022.2	1.0	2023.4	0.0
Reuse-MaTrace model	2018.4	0.4	20224	15	2029.6	0.2
(Extended Data Set)	2010.4	0.4	2022.4	1.5	2020.0	0.2

Table 4.1: Comparison of instant of total in use-stock depletion by models and data sets. The decimal digits represent fractions of years (not months). The deviations are based on the instances when the MaTrace model by Godoy León et al. (2020) is considered. The Combined Data Set consists of data of Godoy León et al. (2020) and Glöser-Chahoud et al. (2019). The Extended Data Set includes additional products for the product category portable batteries (see Chapter 3).

The figure and the table above display, that the deviation of stock-depletion instances regarding the total in-use stock are very small at 75 % and 25 % (less than 0.5 years in both cases). At 50 % the deviation from the MaTrace model by Godoy León et al. (2020) to the Reuse-MaTrace model using Combined Data Set is 1.3 years and to the Reuse-MaTrace model using Extended Data Set 1.5 years.

These findings lead to the conclusion that the modeling changes do not impact the total use-stock in all instances to the same extent. Figure 4.4 A displays the relative stock deviation of the Reuse-MaTrace model using Combined Data Set and Extended Data Set in comparison to the MaTrace model by Godoy León et al. (2020). The orange lines represent the instances when the total in-use stock of the MaTrace model by Godoy León et al. (2020) holds 75 %, 50 %, and 25 %.



Figure 4.4: Relative comparison of the total in-use stock (a) and the portable batteries in-use stock (b) of the MaTrace model by Godoy León et al. (2020) (orange line), the Reuse-MaTrace model using the Combined Data Set (green line, consisting of data of Godoy León et al. (2020) and Glöser-Chahoud et al. (2019)), and the Reuse-MaTrace model using the Extended Data Set (red line, the data set includes additional products for the product category portable batteries, see Chapter 3). The vertical, orange lines mark the instances in time when the stock of the MaTrace model by Godoy León et al. (2020) holds only 75 %, 50 %, and 25 % of the initial inflow (see Figure 4.3).

It can be seen, that both lines peak in the year 2021, which is the instance when the total in-use stock of the MaTrace model by Godoy León et al. (2020) is depleted to 50 %. At this instance, the total in-use stock of the Reuse-MaTrace model using the Combined Data Set is about 13 % larger and when the Extended Data Set is applied it is about 11.5 % larger. Additionally, it can be seen that the total in-use stock of the Reuse-MaTrace model using the Extended Data Set is constantly above 0, while the line for the Combined Data Set goes goes below 0 in the very beginning. Therefore, I concluded that

the total in-use stock over all increased for all instances when the MaTrace model by Godoy León et al. (2020) is extended (the following section will discuss the actual impact of the consideration of multiple use cycles).

Looking at Figure 4.3 B it can be seen that the stock behaviour of the different models is different when only the in-use stock of portable batteries is considered. In contrast to the total in-use stock, all lines are distinguishable from the very beginning and their deviation becomes larger till the point when the in-use stock of portable batteries of the MaTrace model by Godoy León et al. (2020) is almost depleted. Table 4.2 gives the instances when the potable batteries in-use stock of each model has decayed to 75 %, 50 %, and 25 %.

Instant when the	7	5 %	5	0%	2	5 %
in-use stock is depleted to:	Year	Deviation	Year	Deviation	Year	Deviation
MaTrace model by Godoy León et al. (2020)	2017.2	0.0	2018.4	0.0	2019.8	0.0
Reuse-MaTrace model (Combined Data Set)	2016.9	-0.3	2018.4	0.0	2020.5	0.7
Reuse-MaTrace model (Extended Data Set)	2017.4	0.2	2019.0	0.6	2021.1	1.3

Table 4.2: Comparison of instants of portable batteries in-use stock depletion by models and data sets. The decimal digits represent fractions of years (not months). The deviations are based on the instances when the MaTrace model by Godoy León et al. (2020) is considered. The Combined Data Set consists of data of Godoy León et al. (2020) and Glöser-Chahoud et al. (2019). The Extended Data Set includes additional products for the product category portable batteries (see Chapter 3).

The stock of all models reaches 75 % within a instant deviation of 0.5 years. When the stock decays to 25 % it is already at 1.3 years. This difference is also reflected in Figure 4.4 B. It can be observed, that the relative portable batteries in-use stock deviation for both data sets is increasing steeply, after the line considering the Reuse-MaTrace model using the Combined Data Set reaches its minimum in 2017. Although, I am tempted to say, that the portable batteries stock increases almost exponentially when reuse is considered in the model, the underlying reason for this behaviour needs to be understood. After the year 2024 the portable batteries in-use stock of the MaTrace model by Godoy León et al. (2020) is almost 0. Therefore, the values of the other models are many times larger which is why a relative comparison is misleading in this case.

In conclusion I found that the total in-use stock as well as the portable battery in-use stock becomes larger when reuse is considered in the modeling process. The following section illustrates the impact of considering a second and a third use cycle.

4.2.2. Use cycle impact

This subsection presents the results of the impact analysis regarding the consideration of multiple use cycles. As explained in Section 4.2.2, this analysis is based on the Reuse-MaTrace model applying the Extended Data Set. The model was run, once considering only one use-cycle of portable batteries. The results of this run were then used as a benchmark in the following. Then the model run considering two use cycles and finally three use cycles. Those consecutive runs created the results represented here.

Figure 4.5 A displays the total in-use stock and B the in-use stock of portable batteries of the Reuse-MaTrace model using the Extended Data Set considering one, two, and three use-cycles of for products within the *portable batteriers* product category. The vertical lines mark the instances in time when the in use-stock holds 75 %, 50 %, and 25 % of the in-use stock.



Figure 4.5: Comparison of the total in-use stock (a) and the portable batteries in-use stock (b) of the Reuse-MaTrace model when one use cycle (blue line), two use cycles (magenta line), or three use cycles (golden line) are considered. The vertical lines mark the instances in time when the representative stock holds only 75 %, 50 %, and 25 % of the initial inflow.

Focusing on the total in-use stock (Figure 4.5 A) shows, that only the lines for one and three use cycles are clearly distinguishable, since the line representing three use cycles appears to cover the line representing two use cycles. This is also reflected in Table 4.3 which displays the instances when the total in-use stock holds 75 %, 50 %, and 25 %. The instances when three use-cycles are considered are always 0.1 years larger.

Instant when the	7	5 %	5	50%	2	5 %
in-use stock is depleted to:	Year	Deviation	Year	Deviation	Year	Deviation
One use cycle	2018.5	0.0	2021.5	0.0	2029.4	0.0
Two use cycles	2018.9	0.4	2022.2	0.7	2029.5	0.1
Three use cycles	2019.0	0.5	2022.3	0.8	2029.6	0.2

Table 4.3: Comparison of instant of total in use-stock depletion by number of use cycles. The decimal digits represent fractions of years (not months). The deviations are based on the instances when only one use cycle was considered.

The deviation of instances is the largest in the column presenting the instances for 50 % stock depletion. So, it can be assumed that the impact of considering additional use-cycles for *portable batteries* is the largest between the instant when the in-use stock holds 75 % and the instant when it holds 25 %. This assumption gets confirmed by analysing Figure 4.6. The figure displays the relative stock deviation of the results considering two and three use-cycle in comparison to the results when only one use-cycle was considered. Figure 4.6 A displays the deviation of the total in-use stock where the orange lines represent the instances when the total in-use stock holds 75 %, 50 %, and 25 % when only one use cycle is considered.



Figure 4.6: Relative comparison of the total in-use stock (a) and the portable batteries in-use stock (b) when one use cycle (blue line), two use cycles (magenta line), or three use cycles (golden line) are considered. The vertical, blue lines mark the instances in time when the stock of the Reuse-MaTrace model which considers only one use cycle holds only 75 %, 50 %, and 25 % of the initial inflow (see Figure 4.5).

As it can be seen, both lines peak around the instance when the in-use stock is depleted by 50 %. In this peak, the total in-use stock is about 8.5 % larger when three use-cycles are considered and about 7.5 % larger when only two use-cycles are considered. Hence, considering the additional third use-cycle results in an stock increase of 1 % which proofs that it does make a difference even though this was not visible in Figure 4.5.

Focusing on the in-use stock behaviour of the *portable batteries* product category I found an overall similar behaviour as for the previous impact analysis (see Section 4.2.1). Looking at Figure 4.5 B one finds that the lines representing two and three use-cycles deviate early on from the line which represents one use cycle. This deviation increases until the line representing one use-cycle starts to approach 0 % around the year 2022. This behaviour is also reflected in Table 4.4. The table gives the instances when the potable batteries in-use stock of each considered case has decayed to 75 %, 50 %, and 25 %. The deviation between the portable battery stock for the case when one use-cycle and when three use-cycles are considered is the smallest when 75 % of the initial stock is still in use and the largest when only 25 % are left in use.

Instant when the	7	5 %	5	50%	2	5 %
in-use stock is depleted to:	Year	Deviation	Year	Deviation	Year	Deviation
One use cycle	2017.2	0.0	2018.3	0.0	2019.9	0.0
Two use cycles	2017.7	0.5	2019.0	0.7	2021.1	1.2
Three use cycles	2017.8	0.6	2019.0	0.7	2021.3	1.4

 Table 4.4:
 Comparison of instant of portable batteries in-use stock depletion by number of use cycles. The decimal digits

 represent fractions of years (not months).
 The deviations are based on the instances when only one use cycle was considered.

When looking at Figure 4.6 B, the relative deviation of the portable batteries in-use stock behaves similarly to the previous impact analysis (see Section 4.2.1, Figure 4.6). The steep growth in deviation can be explained by the fact, that the in-use stock when one use cycle is considered converges earlier to 0 the in the two other cases.

4.3. Monte Carlo simulations

Section 2.4 discussed that the purpose of the Monte Carlo simulations and the analysis of the results is to find the inputs which insert the greatest uncertainty into the model. This is measured with C (see equation 2.5), the relative contribution of an input to the variance of the considered results. The variance of the results is introduced by the randomization of the inputs before each run. Therefore, it is important to be aware of the variance of the regarded result at each time since the relevance of a high C-value vanishes when the variance in the model is very small. The result I deemed to be the most relevant is the total in-use stock, since this is the most relevant measure regarding the resource efficiency of materials, in this case cobalt.

Figure 4.7 displays the variance of the total in-use stock represented by several confidence bands. Those bands base on quantiles. A quantile is calculated by arranging the results in ascending order (e.g. the in-use stock of all runs in the year 2025). Considering that the Monte Carlo simulations contain 10,000 runs, the 1 % quantile would be the 100th value of those ordered result. Hence, a confidence band covers 98 % of all values between the 1 % quantile and the 99 % quantile. The wider the confidence bands, the larger the variation in the results.



Figure 4.7: Confidence bands of the total in-use stock derived from Monte Carlo simulations (a) and the absolute deviation of the confidence bands from the median of the total in-use stock (b)

As the figure shows, there is no variation in 2015 because all models start with an inflow of 100 %. The confidence bands are especially wide between the years 2018 and 2020 (see Figure 4.7) and become constantly narrower until the variance is very small in the year 2050. This alternation in variance is important to consider in the following course. Besides the variation, the actual in-use stock size is important to consider, if the stock is almost depleted in the year 2050, it is not expediently to discuss which input inserts the most uncertainty into the model at this point in time. For this reason, the following graphs contain a line indicating the total in-use stock (dashed red line) and a line indicating the 1 % to 99 % quantile uncertainty bandwidth (solid purple line). Here the bandwidth is defined as the absolute difference between the 1 % quantile and the 99 % quantile per instance.

Figure 4.8 displays the relative contribution of the inputs to the variance (in the following uncertainty) of the total in-use stock. The uncertainties of the inputs are summed by input vector (e.g. the sum of the contribution of all recycling efficiencies instead of the contribution of recycling efficiency for portable batteries, mobility batteries etc.).



Figure 4.8: Share of uncertainty in total in-use stock by input vectors. The dashed red line represents the total in-use stock and the solid purple line represents the 1 % to 99 % quantile uncertainty bandwidth as measure for the uncertainty present at the regarded instances.

The year 2015 is not represented since no uncertainty is present as explained, which can also be seen in Figure 4.7 B. From 2016 onward, the most important contributor is η the distribution of the initial inflow. Despite its decline it is deemed the most important one since it is especially strong, when the total in-use stock is still high (consider the vertical lines in the graph) and when the variation (solid purple line) in the total in-use stock is largest.

The second most important contributor is the Weibull scale of the in-use stock for all product categories except portable batteries (all product categories where reuse is not considered). The lifetimes of the larger share of the material is determined by this input, thus it is intuitive that it contributes to the uncertainty in the system which also explains its increased importance. Most Weibull shapes β are larger than 1 which means that the survival rate in the first few years is high (also see Figure 2.5). The Weibull scale affects the point in time when the failure rate increases. Therefore, the contribution to the uncertainty of these Weibull scales increase over time. Far less relevant are the next largest contributors, Weibull shapes connected to the mentioned Weibull scales and the split of the material in portable batteries over product δ . The contribution of the Weibull shape only becomes considerably large after 2040, therefore, I will not discussed or considered it any further. At this point in time the largest share of the stock is already depleted and the variation in the results is small in this period (see Figure 4.7). Looking at the data (see Appendix A) it is explainable why δ holds some uncertainty. Products have different lifetimes and transfer coefficients from a use phase to the next use phase and from hibernating stocks to the next use phase. Some products have configurations which contribute more to resource efficiency than others, to me it makes sense to see this reflected in the mapping of uncertainties.

Since the main contributor η is significantly more impact full than all other inputs, it will be investigated more closely. Figure 4.9 displays the uncertainties of the components of the vector η (material share allocated to product category) ignoring all other sources of uncertainty.



Figure 4.9: Share of uncertainty in total in-use stock by the split of the initial input over product categories. The dashed red line represents the total in-use stock and the solid purple line represents the 1 % to 99 % quantile uncertainty bandwidth as measure for the uncertainty present at the regarded instances.

Looking at the graph, some observations can be explained by consulting the input data while some cannot. The most obvious detection is the increasing contribution of the initial share of superalloys to the uncertainty of the total in-use stock. This is the product category with the longest expected lifetime. Therefore, it is understandable that its share introduces uncertainty in the long run when all other stocks are almost depleted. The same applies for the category pet precursors catalysts, the category with the shortest lifetime. Hence, its share of the initial inflow determines the size of the total in-use stock especially in the beginning. Other phenomena cannot be explained. For example the two occurrences of dissipative uses. At this point I have to admit that the model might be too complex to fully comprehend its dynamics and that I cannot fully explored them within the scope of this thesis.

In order to learn more about inputs having a smaller but considerable impact on the total in-use stock, new *C*-values were calculated factoring out the three largest contributes. The results are presented in Figure 4.10.



Figure 4.10: Share of uncertainty in total in use-stock by input vectors excluding the three major uncertainty contributor. Transparent contributors are not listed in the legend since their contribution was deemed too small. The dashed red line represents the total in-use stock and the solid purple line represents the 1 % to 99 % quantile uncertainty bandwidth as measure for the uncertainty present at the regarded instances.

The legend presents the six input vectors which hold the largest uncertainty on average. The split of portable battery material to products was already discussed. The Weibull scale for the first use cycle of portable batteries is the next largest contributor. This is in line with the findings above. The parameters defining the lifetime are important. It is also reflected in the third largest parameter in Figure 4.10,

the share of products containing portable batteries going directly to a second use cycle. This input is important regarding the total lifetime of a product.

5

Discussion

Driven by future resource scarcity and the implications of raw material extraction this research was conducted to extend and to improve an existing MaTrace model by adding a state of the art representation of a circular strategy (see Table 1.1). Reuse was chosen as the strategy which lead to the following main research question.

How to introduce reuse into a MaTrace model and how to evaluate it?

The first sub-research questions concerns the impact of this venture,:

To what extent is the outcome of the MaTrace model as described by Godoy León et al. (2020) altered, when the reuse of consumer electronics is modelled following the approach of Thiébaud et al. (2017)?

The second sub-research question is concerned with the uncertainties to gain an even better understanding about the model behaviour:

Which input parameters introduce the most uncertainty into the Reuse-MaTrace model and what insights can be derived from analysing uncertainties?

The following will discuss to what extent this research can answer those questions. In order to answer the first part of the main research question, the modelling approach and its limitations will be discussed. The used data, assumptions and uncertainties will be reviewed before diving into the model output and impact answering the first sub-research question. In order to answer the second sub-research question, the outcomes of the Monte Carlo simulations will be interpreted. On this basis the second part of the main research question shall be answered.

The final section of this chapter entails recommendations and suggestions for further research.

5.1. Modeling approach and limitations of the model

The presented Reuse-MaTrace model combines two existing models. The novelty of this work is the application of the two models in interrelation to each other. The MaTrace model by Godoy León et al. (2020) describes the effects of the recycling of cobalt in the EU while the reuse model by Thiébaud et al. (2017) adds another circular strategy to the model.

Although, reuse was considered in previous MaTrace models (Klose & Pauliuk, 2021; Pauliuk et al., 2017; Zhang et al., 2019) this is the first model which follows the approach of Thiébaud et al. (2017) considering individual use, reuse and hibernating times for different products and multiple use cycles. Previous MaTrace implementations modeled reuse in a more simplistic way redirecting end-of-life products back to the in-use stock without considering changes in lifetime.

The two used models are very different in nature. The MaTrace model appears to be very technical on first glance. It relies to a large extend on economic and technical input. The data origin of the reuse model is based on the outcome of social scientific research while the model itself follows a typical MFA approach. Therefore, the newly created model builds an important interface between social science (e.g., lifetimes and reuse), economics (e.g., sales numbers and exports), and natural science/engineering (e.g., recycling efficiencies).

As it was discussed in Section 5.3 the inputs originating from social science research namely the lifetime of products containing portable batteries and the share of products being lost due to nonselective collection determine to a large extend the resource efficiency of cobalt. The model has those numbers as inputs but the underlying concepts and mechanisms are not reflected. Chapter 1 describes that a product can become obsolete to the user for many reasons which determines at least the time of a use-cycle if not the entire lifetime. Furthermore, the knowledge and perception towards waste management determines (alongside with other factors) how well households are separating their waste. Those concepts are not reflected in the model.

This critical property limits the usefulness of the Reuse-MaTrace model for decision and policy makers. The necessary model input regarding those crucial factors can be gathered by conducting interviews and surveys. So, it is possible to model a current state. Furthermore, it is possible to develop desired states by applying scenarios with e.g., longer lifetimes or lower non-selective collection rates. The model can present the current state and what needs to be achieved but not how to do so.

Another limitation of the model is its incompleteness assuming the target is to implement all possible strategies as presented in the 9R-framework (see Table 1.1). The first cluster of suggested strategies "Smarter product use and manufacture" holds strategies which impacts can be reflected via the input data (e.g., R2: Reduce - lowering the inflow of new products). However, the strategies in the second cluster "Extend lifespan of product and its parts" concern the fate of a product at its end of life. Thus, those strategies can be integrated into the Reuse-MaTrace model. Figure 5.1 shows a simplified conceptual integration of those strategies into the model.



Figure 5.1: Simplified conceptual integration of 9R strategies (in blue) into the Reuse-MaTrace model

Looking at this depiction it must be kept in mind that the representation is highly simplified and

that integration of each strategy deserves as much or more thought than the integration of reuse. Considering for example refurbishment, it has to be figure out to which use-cycle the refurbished product can be matched. Is it a as good as new product, or may it become obsolete sooner due to one of the various kinds of obsolescence (see Section 1.1.3)? It might also be that a separate in-use stock for refurbished products has to be introduced since feeding the products back to any use-cycle may not be an adequate modeling choice at all. Those kind of considerations have to be discussed in future research.

The following section treats the used data and its inherent uncertainties since the results of the case study and the Monte Carlos simulation depend to a large extend on this.

5.2. Used data and uncertainties

The data of the model originates from three different main sources. The data for the use phase of non-consumer products, the end of life phase and the production phase are taken from Godoy León et al. (2020) who gathered and assessed their data in their previous publication (Godoy León & Dewulf, 2020). The data for the use phase of portable batteries originates from Glöser-Chahoud et al. (2019)). I extended this data set in order to introduce more diversity in product lifetimes (see Chapter 3).

The data quality and uncertainties were already discussed extensively in Section 2.4.2. As it can be derived from the uncertainty scores (see Appendix B) the bigger share of the received scores are larger than 2 which leads to the conclusion that the data quality is overall low. This has important implications for multiple potential stakeholders as modelers who might have an interest to further develop the model or who get inspired. It is important to recognize that this low availability of high quality data is a source of uncertainty beyond the made assumptions and implications of the model itself. Although one might intuitively think that expanding models might bring more precise results this does not necessarily hold true when the additionally required data is as uncertain. Therefore, it is important to critically assess data availability before taking the decision to apply parts of the model or to extend this model.

The knowledge and consideration of the data quality is equally important for model users who might be in the position to communicate the results or who might base decision on the model output. It has to be understood that the output can only be as good as the model itself in interrelation with the data quality. When the model is applied on another use case it appears to be sensible to execute Monte Carlo simulation as it was done in this research. This helps to gain an understanding about the possible magnitude of the divergence between results and reality.

Although, other interested industrial ecologist may not plan to use the model or its results it is important to realize how hard it is to find valid data to describe societies metabolism. Besides the modeling of material flow, research regarding the accounting and quantification is still urgently needed.

5.3. Impact and model output

Looking at the impact of considering multiple use-cycles, it was found that the total in-use stock increases by 8 % when three use cycles are considered instead of one. Even though, this result is only representative for this specific case study, it shows the clear potential of reuse regarding the in resource efficiency.

The general stock dynamics were described in Section 4.1 and several conclusions can be derived from those and from Figure 4.1. There are four stocks which mainly define the total in-use stock: portable batteries, dissipative uses, hard metals, and super alloys. The case of disspative uses was discussed in Section 4.1. Looking at hard metals and super alloys, it became obvious that those stocks deplete the slowest. The underlying reason for this is most likely their long lifetime. Furthermore, it can be seen that the portable battery stock depletes the quickest, even though reuse is considered. This dynamic has sever consequences regarding the total in-use stock and therefore for the resource efficiency. Portable batteries make up the largest share of the original inflow.

Although, the consideration of reuse changes the magnitude of the total in-use stock significantly in a relevant time frame of the considered period, it does not change the overall stock and flow dynamics. The dynamics as described above are inline with the findings of Godoy León et al. (2020).

But the possibility of reusing products is also a symptom of another threat regarding resource efficiency. Consumers discard functioning products due to various kinds of obsolescence as described in Section 1.1.3. Makov and Fitzpatrick (2021) pointed out, that perceived obsolescence is normally driven by non-technical aspects and consumers tent to underestimate the performance of products they

perceive as obsolete.

Since this specific case study treats the reuse of portable batteries, it needs to be considered that portable batteries are components with a rather short lifetime in comparison to other parts of the products. Their expected lifetime lies within three to five years (Beaulieu, 2021). This means that some batteries may have an actual potential to reach their technical end of life while the functional lifetime of batteries also depends on the charging habits of the users.

Another outstanding finding reflected in Figure 4.1 is the distribution of losses and the fact that the largest share of lost material is due to non-selective collection (this finding is as well in line with the findings of Godoy León et al. (2020)). This misplacement of products into the wrong waste collection stream is present for consumer products (portable batteries) but as well for catalysts and hard metals. Hence, the largest loss in the system is not caused by technical limitations but by human error.

The next largest losses are pre-treatment, recycling, and production losses. This is the order in which the processes (pre-treatment, recycling, and production) occur within the system. This circumstance can explain the order of largest losses. By the time the material reaches the production, a large share of it was already lost in the steps before. Therefore the share of production waste is small by default. Therefore, one has to look at the actual efficiencies of the processes since Figure 4.1 might be misleading in that regard. It can be said though, that an increase in recycling and production efficiency can only lead to a relevant gain in overall resource efficiency once less material is lost due to non-selective collection. The overall efficiency of a process chain is calculated via the product of all efficiencies. Hence, all efficiencies have to be high to achieve an overall high efficiency.

Another large share is the accumulated share of exported material. It consist of EOL products, secondary materials, and produced products. The whereabouts of that outflow are not treated by this model.

In summary it can be said, that the consideration of reuse has a significant impact on the total in-use stock. Furthermore, regarding this case study it can be said that further improvements of lifetimes and the recycling performance could be achieved through a change in user behaviour.

5.4. Monte Carlo Simulations

Looking at the outcome of the Monte Carlos simulations regarding the total in-use stock, the main contributor to uncertainty is easily identifiable. Figure 4.8 shows clearly that the split of the initial inflow δ is the input vector which carries the most uncertainty by a large margin. The second most uncertain input vector is the one defining the Weibull scales T for the product categories where products are not reused (all except *portable batteries*). The Weibull scale defines the point in time when 63.2 % of the stock is depleted. Therefore, it determines the survival curve of a product category to a large extent. Considering the article of Miatto et al. (2017) as discussed in Section 1.1.5, it is surprising that Weibull shapes do not contribute much uncertainty in this case study. The shapes of used survival curves are commonly considered to be a source of uncertainty in MFA modelling. The low impact of the shape (in this case) might be due to the variation mechanism used for Weibull shapes. If the initial Weibull shape was larger then 1, the variations considered in the Monte Carlo simulations were limited to values larger then 1. This was done to avoid too large variations in shapes. Hence, this finding may not be generalizable.

Focusing on what can be said based on the results, it became clear by looking at Figure 4.9 that the initial inflow distribution is so important because the lifetimes among the product categories differentiate extremely. The product category with the longest lifetime is super alloys (T = 17.3) and the one with the shortest is pet precursors catalyst (T = 0.57) (see Appendix A). The Weibull scales were varied by ±50 % which means that the longest lifetime could have never become the smallest and vise versa. Hence, the split of the initial inflow δ contributes more to the variance of the result since this has a larger impact on the survival of the total in-use stock then the variation of the survival curves itself.

Comparing the outcome of the Monte Carlo simulations with the results as presented in Section 4.1 and as discussed in Section 5.3 one might expect to see the largest loss due to non-selective collection reflected. This is not the case. The collection-to-recycling rate σ is not even relevant in Figure 4.10 which displays minor contributors to the uncertainty in the in-use stock. There are two possible explanations for this inconsistency in results. Firstly, its possible that the magnitude in variation of the collection-to-recycling rate σ was too low. Looking at Table B.10 in Appendix B one finds that σ is rated with an uncertainty score of 2 for many product categories including portable batteries (holding the

largest share of the initial stock). σ is not varied very strongly due to this low uncertainty. Therefore, it is possible that other inputs had a larger impact on the total in-use stock which is why no correlation was found. Secondly, it could be that there is indeed no strong correlation. As explained in Section 5.3 it might be that a significant share of material is lost in the recycling process once a better waste separation is achieved. Hence, it is possible that a local improvement does not advance the overall performance of the system significantly.

The target of the Monte Carlo simulations was to learn more about the uncertainties of the model itself. The approach taken was only successful to a limited extend since the largest reveled uncertainties are still connected to the initial input data. This insight is not surprising considering the used approach. The inputs were varied based on the estimated uncertainty in the input data. An variation mechanism independent from the data quality might have revealed phenomena closer connected to the model itself instead of the data. Furthermore, the consideration of other distributions to model survival curves may have given interesting insights.

Furthermore, it would have been interesting to see the extension of the MaTrace model reflected in the results. However, it is not necessarily bad that input parameters to the reuse part of the model were not contributing much uncertainty. It could be prove that the extension by the reuse model changes the results significantly (see Section above and Section 4.2.2). Therefore, it can be concluded that the model extension adds detail without increasing the uncertainty significantly.

Another very important learning can be derived from the results of the Monte Carlo simulations: The distribution of the material over the different product categories is more sensitive than the survival curves of these categories. This has implications for different stakeholders. If modelers extend this model they have to be aware that this input needs a lot of attention to determine the shares as accurate as possible. Same holds true for model users applying the model to another context. In both cases it is recommended to execute Monte Carlo simulations because as described, the uncertainties revealed depend to an extend on the input data. Focusing on this specific case study one realizes that an effective way to improve resource efficiency is to decrease the material inflow into product categories with short lifetimes (e.g. pet precursors catalyst or portable batteries).

5.5. Recommendations

On the basis of the created model, the results, and the discussion multiple recommendations for further research can be derived. To begin with, the model behaviour is not sufficiently explored yet although Monte Carlo simulations were conducted. Therefore, I recommend to apply this model on other materials and other geographic regions to test the model with diverse data sets.

Throughout this work, data and data quality played an important role. By scoring the data it became clear that a large share of the inputs are rather uncertain. Therefore, more research and more data collection has to be conducted to obtain high quality data.

Looking at the model itself, there are multiple possibilities to further extend the model. As suggested in Section 5.1, the integration of the 9R framework into the model could be pushed by implementing the strategies repair, refurbish, remanufacturing, or repurpose. A possible starting point to integrate remanufacturing could be the work of Zhang et al. (2021) which integrated the reuse of old vehicle engines in new vehicles into their MaTrace model.

The integration of reuse adds a dimension of consumer behaviour to the model. Although, this work may be a step in the right direction it does not consider consumer behaviour sufficiently. Regarding the results of the case study, a reoccurring finding is the potential influence of consumer behaviour. Consumer behaviour influences the distribution of the initial inflow, lifetimes, hording times, reuse rates, and waste separation - all key aspects regarding resource efficiency. Therefore, more research on how to represent consumer behaviour in MFA models needs to be conducted.

Looking at the overall model output as presented in Figure 4.1, I found that exports make up a large share in the long run. The whereabouts of the exports are admittedly a blind spot of this model. A model solving this problem is MaTrace Global by Pauliuk et al. (2017). Therefore, combining the concepts of Reuse-MaTrace and MaTrace Global might offer insights when investigating the impact of circular strategies on a global scale.

The fact that MaTrace models generally only consider the inflow of one cohort is an obvious weakness. Although, MaTrace models depend on dynamic stock modelling they only consider the inflow of this single year while leaving other parameters as transfer coefficients and efficiencies static. This

applies to this thesis as well. As MFA evolved from static to dynamic models, this process has to be repeated for MaTrace models. Therefore, I recommend to create MaTrace models using time series data, both for inflows and other model parameters.

6

Conclusion

The conviction that circular strategies have a positive impact on resource efficiency led to the research objective; integrating a compelling reuse model into a MaTrace model. Reuse has been integrated into MaTrace models before, but as described only in simplistic ways. Therefore, the research question asks: *How to introduce reuse into a MaTrace model and how to evaluate it?*

The first part of this research question was answered methodologically: A state of the art reuse model was found and integrated into the MaTrace model. Consumer goods present in the original MaTrace model are portable batteries. The inflow into this product category was redirected into the reuse model. There the inflow was split into a diverse group of consumer electronics which passed through the reuse model. The outflow, the end of life products, of the reuse model were then fed back into the original MaTrace model which represents the end of life treatment, recycling, and production. The first sub-research question asked about the impact of the consideration of reuse. It was found that the total in-use stock increases by 8 % in the peak when reuse is considered. Hence, it was concluded that the consideration of reuse can have a significant impact on the modeled in-use stock. Beside this, the gross stock dynamics remained intact. The overall findings are in line with the results of Godoy León et al. (2020), the creator of the used MaTrace model.

The second part of the main research question on the evaluation of the model also inspired the second sub-research question. It asked which inputs introduce the most uncertainty into the created Reuse-MaTrace model. On this basis Monte Carlo simulations were conducted in order to gain insights regarding the model behaviour. It was evaluated which model inputs have the largest impact on the total in-use stock, since this was deemed the relevant measure regarding resource efficiency. An assigned input uncertainty score defined how the inputs were varied. The results showed that the by far most influential input parameter is the split of the inflow. This is due to the strongly diverging lifetimes among product categories. This was followed by the lifetimes themselves.

The results of the conducted Monte Carlo simulations gave valuable insights since it identified the most sensitive model inputs. However, the results appear to be strongly connected to the input data and can be hardly generalized to describe the pure model behaviour. A more rigorous randomization mechanism detached from data uncertainty may have served this purpose better.

On the basis of this work multiple ideas for further research came to mind. To start with, this work added only one circular strategy to the MaTrace model. Other circular strategies can be implemented to obtain a more complete representation of a circular economy.

Drawing a parallel of between the evolution of MaTrace models and the development of MFA in recent decades, MaTrace models have to become dynamic. Therefore, I suggest the development of MaTrace models using time series data for inflows and other model parameters.

Lastly, the integration of the reuse model added a dimension of consumer behaviour to MaTrace. The results of the case study revealed that consumer behaviour also influences other parts of the model as the waste treatment process. I recommend to find better ways to consider and represent consumer

behaviour in MaTrace models. That way the interface between MFA modelling and social science can grow and strengthen the interdisciplinarity in modelling society's metabolism.

References

- B. Müller, D. (2006). Stock dynamics for forecasting material flows—Case study for housing in The Netherlands. *Ecological Economics*, 59(1), 142–156. https://doi.org/10.1016/j.ecolecon.2005. 09.025
- Baccini, P., & Brunner, P. H. (1991). *Metabolism of the anthroposphere* [Section: xii, 157 pages : illustrations; 24 cm]. Springer-Verlag.
- Balton, J. (2022). Bike Statistics & Facts [Of 2022]. https://www.bicycle-guider.com/bike-facts-stats/
- Beaulieu, L. (2021). Phone Battery Lifespan: How Long Does Your Phone Battery Last? Retrieved July 18, 2022, from http://recy-cell.ca/en/how-long-does-phone-battery-last/
- Bobba, S., Mathieux, F., & Blengini, G. A. (2019). How will second-use of batteries affect stocks and flows in the EU? A model for traction Li-ion batteries. *Resources, Conservation and Recycling*, 145, 279–291. https://doi.org/10.1016/j.resconrec.2019.02.022
- Box, J. M. (1983). Extending Product Lifetime: Prospects and Opportunities [Publisher: MCB UP Ltd]. *European Journal of Marketing*, 17(4), 34–49. https://doi.org/10.1108/EUM000000004830
- Burns, B. (2010). Re-evaluating Obsolescence and Planning for It [Num Pages: 22]. *Longer Lasting Products*. Routledge.
- Cai, Y., Xu, W., Ji, D., Yang, Z., & Fu, Q. (2022). Material flow analysis of the nitrogen loading to surface water of Miyun reservoir watershed under uncertainty. *Journal of Cleaner Production*, 353, 131574. https://doi.org/10.1016/j.jclepro.2022.131574
- Calvo, G., Valero, A., & Valero, A. (2017). Assessing maximum production peak and resource availability of non-fuel mineral resources: Analyzing the influence of extractable global resources. *Resources, Conservation and Recycling*, 125, 208–217. https://doi.org/10.1016/j.resconrec. 2017.06.009
- Cao, Z., Shen, L., Zhong, S., Liu, L., Kong, H., & Sun, Y. (2018). A Probabilistic Dynamic Material Flow Analysis Model for Chinese Urban Housing Stock: A Probabilistic Dynamic MFA Model. *Journal* of Industrial Ecology, 22(2), 377–391. https://doi.org/10.1111/jiec.12579
- Chung, S.-s., Lau, K.-y., & Zhang, C. (2011). Generation of and control measures for, e-waste in Hong Kong. Waste Management, 31(3), 544–554. https://doi.org/10.1016/j.wasman.2010.10.003
- Circular economy: Definition, importance and benefits | News | European Parliament. (2015). Retrieved July 21, 2022, from https://www.europarl.europa.eu/news/en/headlines/economy/20151201ST 005603/circular-economy-definition-importance-and-benefits
- den Hollander, M. C., Bakker, C. A., & Hultink, E. J. (2017). Product Design in a Circular Economy: Development of a Typology of Key Concepts and Terms. *Journal of Industrial Ecology*, 21(3), 517–525. https://doi.org/10.1111/jiec.12610
- Desai, P. (2020). Cobalt demand for 5G technology to challenge electric vehicles. *Reuters*. Retrieved March 25, 2022, from https://www.reuters.com/article/us-cobalt-5g-electric-idCAKCN26C1EQ
- Dimov, I. (2008). Monte Carlo methods for applied scientists. World Scientific.
- Duc, P. M., Tuoi, N. T., Hung, C. M., Tuyen, T. M., Quan, D. A., Phuong, V. V., & Anh, D. N. (2022). Industrial Ecology and Eco-Industrial Park: Models and Statistics with Implications to a Central Province in Vietnam [Number: 3]. JOURNAL OF ALGEBRAIC STATISTICS, 13(3), 338–354. Retrieved August 18, 2022, from https://publishoa.com/index.php/journal/article/view/615
- Dunn, J., Slattery, M., Kendall, A., Ambrose, H., & Shen, S. (2021). Circularity of Lithium-Ion Battery Materials in Electric Vehicles [Publisher: American Chemical Society]. *Environmental Science* & Technology, 55(8), 5189–5198. https://doi.org/10.1021/acs.est.0c07030
- Farchy, J., & Warren, H. (2018). China Has a Secret Weapon in the Race to Dominate Electric Cars. *Bloomberg.com.* Retrieved March 22, 2022, from https://www.bloomberg.com/graphics/2018china-cobalt/
- Fishman, T., Myers, R., Rios, O., & Graedel, T. (2018). Implications of Emerging Vehicle Technologies on Rare Earth Supply and Demand in the United States. *Resources*, 7(1), 9. https://doi.org/10. 3390/resources7010009

- Fyhri, A., & Beate Sundfør, H. (2020). Do people who buy e-bikes cycle more? *Transportation Research Part D: Transport and Environment*, *86*, 102422. https://doi.org/10.1016/j.trd.2020.102422
- Gauthier, T. (2001). Detecting Trends Using Spearman's Rank Correlation Coefficient. *Environmental Forensics*, *2*(4), 359–362. https://doi.org/10.1006/enfo.2001.0061
- Glöser, S., Soulier, M., & Tercero Espinoza, L. A. (2013). Dynamic Analysis of Global Copper Flows. Global Stocks, Postconsumer Material Flows, Recycling Indicators, and Uncertainty Evaluation. *Environmental Science & Technology*, 47(12), 6564–6572. https://doi.org/10.1021/es400069b
- Glöser-Chahoud, S., Pfaff, M., Walz, R., & Schultmann, F. (2019). Simulating the service lifetimes and storage phases of consumer electronics in Europe with a cascade stock and flow model. *Journal of Cleaner Production*, *213*, 1313–1321. https://doi.org/10.1016/j.jclepro.2018.12.244
- Godoy León, M. F., Blengini, G. A., & Dewulf, J. (2020). Cobalt in end-of-life products in the EU, where does it end up? - The MaTrace approach. *Resources, Conservation and Recycling*, 158, 104842. https://doi.org/10.1016/j.resconrec.2020.104842
- Godoy León, M. F., Blengini, G. A., Matos, C. T., & Dewulf, J. (2022). Long-term retrospective analysis of the societal metabolism of cobalt in the European Union. *Journal of Cleaner Production*, 338, 130437. https://doi.org/10.1016/j.jclepro.2022.130437
- Godoy León, M. F., & Dewulf, J. (2020). Data quality assessment framework for critical raw materials. The case of cobalt. *Resources, Conservation and Recycling, 157*, 104564. https://doi.org/10. 1016/j.resconrec.2019.104564
- Gottschalk, F., Scholz, R. W., & Nowack, B. (2010). Probabilistic material flow modeling for assessing the environmental exposure to compounds: Methodology and an application to engineered nano-TiO2 particles. *Environmental Modelling & Software*, *25*(3), 320–332. https://doi.org/10. 1016/j.envsoft.2009.08.011
- Guo, X., Zhang, J., & Tian, Q. (2021). Modeling the potential impact of future lithium recycling on lithium demand in China: A dynamic SFA approach. *Renewable and Sustainable Energy Reviews*, 137, 110461. https://doi.org/10.1016/j.rser.2020.110461
- Helbig, C., Kondo, Y., & Nakamura, S. (2022). Simultaneously tracing the fate of seven metals at a global level with MaTrace-multi. *Journal of Industrial Ecology*. https://doi.org/10.1111/jiec.13219
- Hirschnitz-Garbers, M., Montevecchi, F., & Martinuzzi, A. (2013). Resource Efficiency. In S. O. Idowu, N. Capaldi, L. Zu, & A. D. Gupta (Eds.), *Encyclopedia of Corporate Social Responsibility* (pp. 2018–2018). Springer. https://doi.org/10.1007/978-3-642-28036-8 728
- Ianchenko, A., Simonen, K., & Barnes, C. (2020). Residential Building Lifespan and Community Turnover [Publisher: American Society of Civil Engineers]. *Journal of Architectural Engineering*, 26(3), 04020026. https://doi.org/10.1061/(ASCE)AE.1943-5568.0000401
- Jarrín Jácome, G., Godoy León, M. F., Alvarenga, R. A. F., & Dewulf, J. (2021). Tracking the Fate of Aluminium in the EU Using the MaTrace Model [Number: 7 Publisher: Multidisciplinary Digital Publishing Institute]. *Resources*, *10*(7), 72. https://doi.org/10.3390/resources10070072
- Kamran, M., Raugei, M., & Hutchinson, A. (2021). A dynamic material flow analysis of lithium-ion battery metals for electric vehicles and grid storage in the UK: Assessing the impact of shared mobility and end-of-life strategies. *Resources, Conservation and Recycling, 167*, 105412. https://doi.org/10.1016/j.resconrec.2021.105412
- Kastanaki, E., & Giannis, A. (2022). Forecasting quantities of critical raw materials in obsolete feature and smart phones in Greece: A path to circular economy. *Journal of Environmental Management*, 307, 114566. https://doi.org/10.1016/j.jenvman.2022.114566
- Kirchherr, J., Reike, D., & Hekkert, M. (2017). Conceptualizing the circular economy: An analysis of 114 definitions. *Resources, Conservation and Recycling*, 127, 221–232. https://doi.org/10.1016/j. resconrec.2017.09.005
- Kızılersü, A., Kreer, M., & Thomas, A. W. (2018). The Weibull distribution. *Significance*, *15*(2), 10–11. https://doi.org/10.1111/j.1740-9713.2018.01123.x
- Klose, S., & Pauliuk, S. (2021). Quantifying longevity and circularity of copper for different resource efficiency policies at the material and product levels. *Journal of Industrial Ecology*, 25(4), 979– 993. https://doi.org/10.1111/jiec.13092
- Kwarteng, M., Pilík, M., & Juřičková, E. (2018). Beyond cost saving. Other factor consideration in online purchases of used electronic goods: A conjoint analysis approach. *Management & Marketing*, 13, 1051–1063. https://doi.org/10.2478/mmcks-2018-0022

- Lauinger, D., Billy, R. G., Vásquez, F., & Müller, D. B. (2021). A general framework for stock dynamics of populations and built and natural environments. *Journal of Industrial Ecology*, 25(5), 1136–1146. https://doi.org/10.1111/jiec.13117
- List of the verified oldest people [Page Version ID: 1088435497]. (2022). Retrieved May 18, 2022, from https://en.wikipedia.org/w/index.php?title=List_of_the_verified_oldest_people&oldid= 1088435497
- Liu, W., Liu, Y., Liu, Y., Ogunmoroti, A. E., Li, M., Bi, M., & Cui, Z. (2021). Dynamic material flow analysis of critical metals for lithium-ion battery system in China from 2000–2018. *Resources, Conservation and Recycling*, 164, 105122. https://doi.org/10.1016/j.resconrec.2020.105122
- Makov, T., & Fitzpatrick, C. (2021). Is repairability enough? big data insights into smartphone obsolescence and consumer interest in repair. *Journal of Cleaner Production*, *313*, 127561. https://doi.org/10.1016/j.jclepro.2021.127561
- McFarland, M. (n.d.). The next holy grail for EVs: Batteries free of nickel and cobalt. Retrieved August 10, 2022, from https://www.cnn.com/2022/06/01/cars/tesla-lfp-battery/index.html
- Miatto, A., Schandl, H., & Tanikawa, H. (2017). How important are realistic building lifespan assumptions for material stock and demolition waste accounts? *Resources, Conservation and Recycling*, 122, 143–154. https://doi.org/10.1016/j.resconrec.2017.01.015
- Milovantseva, N., & Saphores, J.-D. (2012). Time bomb or hidden treasure? Characteristics of junk TVs and of the US households who store them. *Waste management (New York, N.Y.), 33.* https://doi.org/10.1016/j.wasman.2012.07.020
- Montangero, A., & Belevi, H. (2008). An approach to optimise nutrient management in environmental sanitation systems despite limited data. *Journal of Environmental Management*, *88*(4), 1538–1551. https://doi.org/10.1016/j.jenvman.2007.07.033
- Mudgal, S., Le Guern, Y., Tinetti, B., Chanoine, A., Pahal, S., & Witte, F. (2011). Comparative Life-Cycle Assessment of nickelcadmium (NiCd) batteries used in Cordless Power Tools (CPTs) vs. their alternatives nickel-metal hydride (NiMH) and lithium-ion (Li-ion) batteries (tech. rep.). European Commission – DG ENV. Retrieved April 26, 2022, from https://op.europa.eu/en/publicationdetail/-/publication/eee581bb-25af-46ec-824f-ff78024a7bd7
- Müller, E., Hilty, L. M., Widmer, R., Schluep, M., & Faulstich, M. (2014). Modeling Metal Stocks and Flows: A Review of Dynamic Material Flow Analysis Methods. *Environmental Science & Technology*, 48(4), 2102–2113. https://doi.org/10.1021/es403506a
- Nakamura, S., Kondo, Y., Nakajima, K., Ohno, H., & Pauliuk, S. (2017). Quantifying Recycling and Losses of Cr and Ni in Steel Throughout Multiple Life Cycles Using MaTrace-Alloy. *Environmental Science and Technology*, *51*(17), 9469–9476. https://doi.org/10.1021/acs.est.7b01683
- Nakamura, S., Kondo, Y., Kagawa, S., Matsubae, K., Nakajima, K., & Nagasaka, T. (2014). MaTrace: Tracing the Fate of Materials over Time and Across Products in Open-Loop Recycling [Publisher: American Chemical Society]. *Environmental Science & Technology*, *48*(13), 7207–7214. https: //doi.org/10.1021/es500820h
- Oswald, I., & Reller, A. (2011). E-Waste: A Story of Trashing, Trading, and Valuable Resources. *GAIA* -*Ecological Perspectives for Science and Society*, *20*(1), 41–47. https://doi.org/10.14512/gaia. 20.1.9
- Parajuly, K., Habib, K., & Liu, G. (2017). Waste electrical and electronic equipment (WEEE) in Denmark: Flows, quantities and management. *Resources, Conservation and Recycling*, *123*, 85–92. https://doi.org/10.1016/j.resconrec.2016.08.004
- Pauliuk, S., Kondo, Y., Nakamura, S., & Nakajima, K. (2017). Regional distribution and losses of end-of-life steel throughout multiple product life cycles-Insights from the global multiregional MaTrace model. *Resources, Conservation, and Recycling, 116*, 84–93. https://doi.org/10.1016/j. resconrec.2016.09.029
- Pehlken, A., Albach, S., & Vogt, T. (2017). Is there a resource constraint related to lithium ion batteries in cars? *The International Journal of Life Cycle Assessment*, *22*(1), 40–53. https://doi.org/10. 1007/s11367-015-0925-4
- Pillot, C. (2012). The worldwide battery market 2011-2025. http://kigeit.org.pl/FTP/PRCIP/Literatura/ 063_The_worldwide_battery_market_2011_2025.pdf
- Polák, M., & Drápalová, L. (2012). Estimation of end of life mobile phones generation: The case study of the Czech Republic. Waste Management (New York, N.Y.), 32(8), 1583–1591. https: //doi.org/10.1016/j.wasman.2012.03.028

- Pommeret, A., Ricci, F., & Schubert, K. (2022). Critical raw materials for the energy transition. *European Economic Review*, *141*, 103991. https://doi.org/10.1016/j.euroecorev.2021.103991
- Powell, J. C., Pearce, D. W., & Craighill, A. L. (1997). Approaches to valuation in LCA impact assessment. *The International Journal of Life Cycle Assessment*, 2(1), 11–15. https://doi.org/10.1007/ BF02978709
- Rachidi, N., Nwaila, G., Zhang, S., Bourdeau, J., & Ghorbani, Y. (2021). Assessing cobalt supply sustainability through production forecasting and implications for green energy policies. *Resources Policy*, 74. https://doi.org/10.1016/j.resourpol.2021.102423
- Saphores, J.-D. M., Nixon, H., Ogunseitan, O. A., & Shapiro, A. A. (2009). How much e-waste is there in US basements and attics? Results from a national survey. *Journal of Environmental Management*, *90*(11), 3322–3331. https://doi.org/10.1016/j.jenvman.2009.05.008
- Sartori, I., Sandberg, N. H., & Brattebø, H. (2016). Dynamic building stock modelling: General algorithm and exemplification for Norway. *Energy and Buildings*, *132*, 13–25. https://doi.org/10.1016/j. enbuild.2016.05.098
- SciPy. (2022). Scipy.stats.skewnorm SciPy v1.8.1 Manual. Retrieved July 8, 2022, from https: //docs.scipy.org/doc/scipy/reference/generated/scipy.stats.skewnorm.html
- Sonnemann, G., Tsang, M., Castells, F., & Schumacher, M. (2003). *Integrated Life-Cycle and Risk* Assessment for Industrial Processes. CRC Press. https://doi.org/10.1201/9780203488171
- Sovacool, B. K. (2019). The precarious political economy of cobalt: Balancing prosperity, poverty, and brutality in artisanal and industrial mining in the Democratic Republic of the Congo. *The Extractive Industries and Society*, *6*(3), 915–939. https://doi.org/10.1016/j.exis.2019.05.018
- Steubing, B., Böni, H., Schluep, M., Silva, U., & Ludwig, C. (2009). Assessing computer waste generation in Chile using material flow analysis. Waste management (New York, N.Y.), 30, 473–82. https: //doi.org/10.1016/j.wasman.2009.09.007
- Sun, X., Hao, H., Liu, Z., Zhao, F., & Song, J. (2019). Tracing global cobalt flow: 1995–2015. *Resources, Conservation and Recycling*, 149, 45–55. https://doi.org/10.1016/j.resconrec.2019.05.009
- Takeyama, K., Ohno, H., Matsubae, K., Nakajima, K., Kondo, Y., & Nagasaka, T. (2016). Dynamic material flow analysis of nickel and chromium associated with steel materials by using matrace. *Materiaux et Techniques*, 104(6-7). https://doi.org/10.1051/mattech/2017012
- Tang, C., Sprecher, B., Tukker, A., & Mogollón, J. M. (2021). The impact of climate policy implementation on lithium, cobalt and nickel demand: The case of the Dutch automotive sector up to 2040. *Resources Policy*, 74, 102351. https://doi.org/10.1016/j.resourpol.2021.102351
- Thiébaud, E., Hilty, L., Schluep, M., Böni, H., & Faulstich, M. (2018). Where Do Our Resources Go? Indium, Neodymium, and Gold Flows Connected to the Use of Electronic Equipment in Switzerland. Sustainability, 10(8), 2658. https://doi.org/10.3390/su10082658
- Thiébaud, E., Hilty, L. M., Schluep, M., & Faulstich, M. (2017). Use, Storage, and Disposal of Electronic Equipment in Switzerland [Publisher: American Chemical Society]. *Environmental Science & Technology*, *51*(8), 4494–4502. https://doi.org/10.1021/acs.est.6b06336
- Thiébaud (-Müller), E., Hilty, L. M., Schluep, M., Widmer, R., & Faulstich, M. (2017). Service Lifetime, Storage Time, and Disposal Pathways of Electronic Equipment: A Swiss Case Study [_eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1111/jiec.12551]. *Journal of Industrial Ecology*, 22(1), 196–208. https://doi.org/10.1111/jiec.12551
- Thiébaud (-Müller), É., Hilty, L. M., Schluep, M., Widmer, R., & Faulstich, M. (2018). Service Lifetime, Storage Time, and Disposal Pathways of Electronic Equipment: A Swiss Case Study [_eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1111/jiec.12551]. *Journal of Industrial Ecology*, *22*(1), 196–208. https://doi.org/10.1111/jiec.12551
- Uhunamure, S. E., Nethengwe, N. S., Shale, K., Mudau, V., & Mokgoebo, M. (2021). Appraisal of Households' Knowledge and Perception towards E-Waste Management in Limpopo Province, South Africa. *Recycling*, 6(2), 39. https://doi.org/10.3390/recycling6020039
- Union, E. (2006). Directive 2006/66/EC of the European Parliament and of the Council of 6 September 2006 on batteries and accumulators and waste batteries and accumulators and repealing Directive 91/157/EEC (Text with EEA relevance) (tech. rep.). Retrieved April 26, 2022, from http://data.europa.eu/eli/dir/2006/66/oj/eng
- van Berkel, R., Willems, E., & Lafleur, M. (1997). Development of an industrial ecology toolbox for the introduction of industrial ecology in enterprises—I. *Journal of Cleaner Production*, *5*(1), 11–25. https://doi.org/10.1016/S0959-6526(97)00004-8

- Wang, Y., & Ge, J. (2020). Potential of urban cobalt mines in China: An estimation of dynamic material flow from 2007 to 2016. *Resources, Conservation and Recycling*, 161, 104955. https://doi.org/ 10.1016/j.resconrec.2020.104955
- Weisstein, E. W. (2022). Geometric Series [Publisher: Wolfram Research, Inc.]. Retrieved May 13, 2022, from https://mathworld.wolfram.com/
- Wieser, H., & Tröger, N. (2015). Die Nutzungsdauer und Obsoleszenz von Gebrauchsgütern im Zeitalter der Beschleunigung: Eine empirische Untersuchung in österreichischen Haushalten. Retrieved May 17, 2022, from https://www.research.manchester.ac.uk/portal/en/publications/die-nutzung sdauer-und-obsoleszenz-von-gebrauchsgutern-im-zeitalter-der-beschleunigung(a0ec1d5f-00dc-454a-8b4a-dabd17ee0bee)/export.html
- Williams, E., Hatanaka, T., & Securities, M. (2005). Residential Computer Usage Patterns, Reuse and Life Cycle Energy Consumption in Japan, 12.
- Wilson, G. T., Smalley, G., Suckling, J. R., Lilley, D., Lee, J., & Mawle, R. (2017). The hibernating mobile phone: Dead storage as a barrier to efficient electronic waste recovery. *Waste Management*, 60, 521–533. https://doi.org/10.1016/j.wasman.2016.12.023
- Yoshida, A., Tasaki, T., & Terazono, A. (2009). Material flow analysis of used personal computers in Japan. *Waste Management*, 29(5), 1602–1614. https://doi.org/10.1016/j.wasman.2008.10.021
- Yu, B., An, R., & Zhao, G. (2020). Spatial and temporal disparity of the in-use steel stock for China. *Resources, Conservation and Recycling*, 155, 104667. https://doi.org/10.1016/j.resconrec. 2019.104667
- Zhang, Z., Matsubae, K., & Nakajima, K. (2021). Impact of remanufacturing on the reduction of metal losses through the life cycles of vehicle engines. *Resources, Conservation and Recycling*, 170. https://doi.org/10.1016/j.resconrec.2021.105614
- Zhang, Z., Takeyama, K., Ohno, H., Matsubae, K., Nakajima, K., & Nagasaka, T. (2019). An estimation of the amount of dissipated alloy elements in special steel from automobile recycling. *Materiaux* et Techniques, 107(1). https://doi.org/10.1051/mattech/2019007

Model input data

A.1. Use phase

Product categories	Initial inflow split $arepsilon$	Weibull shape β	Weibull scale T
portable batteries	0.412		
mobility batteries	0.012	2.2	10.63150071
hydroprocessing catalysts coke	0.020	2.5	2.894741746
hydroprocessing catalysts poisoning	0.000	2.5	1.505265708
hydroformylation catalysts	0.003	2.5	2.315793397
pet precursors catalysts	0.017	2.5	0.578948349
dissipative uses	0.090	3.5	14.43516632
hard metals	0.230	1.16	8.915234823
magnets	0.006	1.93	13.90502246
other metallic uses	0.002	1.47	14.62806396
superalloys	0.208	1.74	17.28254338

Table A.1: Inputs initial flow split and Weibull parameters for all product categories except for portable batteries

Products	Hoarding rate	Hoarding time (years)
portable batteries	0.520	4
mobility batteries	0.000	0
hydroprocessing catalysts coke	0.000	0
hydroprocessing catalysts poisoning	0.000	0
hydroformylation catalysts	0.000	0
pet precursors catalysts	0.000	0
dissipative uses	0.000	0
hard metals	0.670	1
magnets	0.500	5
other metallic uses	0.500	5
superalloys	1.000	5

Table A.2: Inputs for hibernating stock of all product categories except for portable batteries

Products	Split portable batteries to products δ
smartphones	0.185
mobile phones	0.018
tablets	0.173
laptops	0.350
e-bikes	0.063
power tools	0.041
others	0.170

Table A.3: Inputs split portable batteries to products

Products	Weibull scale T	Weibull shape β	to use α	to storage β
smartphones	2.500	1.700	0.400	0.400
mobile phones	2.000	1.600	0.400	0.500
tablets	4.000	2.000	0.300	0.400
laptops	5.000	2.000	0.200	0.400
e-bikes	4.614	3.000	0.070	0.220
power tools	9.970	1.500	0.050	0.150
others	3.270	2.130	0.210	0.460

Table A.4: Inputs reuse first life-cycle

Products	Weibull scale T	Weibull shape β	to use γ
smartphones	2.500	1.700	0.500
mobile phones	3.500	1.900	0.500
tablets	3.000	1.800	0.300
laptops	3.000	1.800	0.300
e-bikes	2.242	1.700	0.140
power tools	3.362	1.700	0.100
others	2.160	1.170	0.060

Table A.5: Inputs hibernating stock after first use-cycle

Products	Weibull scale T	Weibull shape β	to use α	to storage β
smartphones	2.000	1.600	0.200	0.200
mobile phones	2.000	1.600	0.200	0.200
tablets	2.000	1.600	0.100	0.300
laptops	2.000	1.600	0.100	0.300
e-bikes	4.643	2.500	0.030	0.260
power tools	5.315	1.200	0.020	0.200
others	3.590	2.170	0.080	0.650

Table A.6: Inputs reuse second life-cycle

Products	Weibull scale T	Weibull shape β	to use γ
smartphones	2.000	1.600	0.200
mobile phones	2.000	1.600	0.200
tablets	2.000	1.600	0.300
laptops	2.000	1.600	0.300
e-bikes	1.672	1.590	0.700
power tools	2.229	1.590	0.050
others	3.900	1.460	0.190

Table A.7: Inputs hibernating stock after second use-cycle

Products	Weibull scale T	Weibull shape β	to storage β
smartphones	1.5	1.6	0.3
mobile phones	1.5	1.6	0.3
tablets	2	1.6	0.4
laptops	2	1.6	0.4
e-bikes	2.215464335	1.5	0.29
power tools	2.072726825	1.1	0.3
others	3.59	2.17	0.7

Table A.8: Inputs reuse third life-cycle

Products	Weibull scale T	Weibull shape β
smartphones	2.000	1.600
mobile phones	2.000	1.600
tablets	1.000	1.500
laptops	1.000	1.500
e-bikes	1.672	1.590
power tools	2.229	1.590
others	3.900	1.460

Table A.9: Inputs hibernating stock after third use-cycle

A.2. End-of-life phase

.

Product categories	Fraction of collected EOL products exported ψ_{EOL}	Collection to recycling rate σ	Pre-treatment efficiency λ_{PT}
portable batteries	0.200	0.640	0.950
mobility batteries	0.000	1.000	0.850
hydroprocessing catalysts coke	0.000	1.000	0.850
hydroprocessing catalysts poisoning	0.000	0.000	0.000
hydroformylation catalysts	0.000	0.900	0.850
pet precursors catalysts	0.000	0.500	0.850
dissipative uses	0.000	0.000	0.000
hard metals	0.000	0.470	0.770
magnets	0.000	0.990	0.650
other metallic uses	0.000	0.840	0.650
superalloys	0.300	0.810	0.650

Table A.10: Inputs collection and pre-treatment

Product categories	Chemical	Zn	Downcycling
portable batteries	1.000	0.000	0.000
mobility batteries	1.000	0.000	0.000
hydroprocessing catalysts coke	0.500	0.000	0.500
hydroprocessing catalysts poisoning	0.000	0.000	1.000
hydroformylation catalysts	1.000	0.000	0.000
pet precursors catalysts	0.000	0.000	1.000
dissipative uses	0.000	0.000	0.000
hard metals	0.630	0.370	0.000
magnets	0.000	0.000	1.000
other metallic uses	0.000	0.000	1.000
superalloys	0.610	0.000	0.390
Recycling efficiency	0.897	0.950	

Table A.11: Inputs distribution to recycling processes	s B and recycling efficiency
--	------------------------------

Product categories	Co metal or compound	W-Co powder
portable batteries	0.000	0.000
mobility batteries	0.021	0.000
hydroprocessing catalysts coke	0.014	0.000
hydroprocessing catalysts poisoning	0.014	0.000
hydroformylation catalysts	0.005	0.000
pet precursors catalysts	0.024	0.000
dissipative uses	0.167	0.000
hard metals	0.313	1.000
magnets	0.017	0.000
other metallic uses	0.003	0.000
superalloys	0.423	0.000
Fraction of exported secondary material \$\psi_{R}	0.133	0.000

Table A.12: Inputs distribution of secondary material and fraction of exported secondary material

A.3. Production

			Processing	Manufacturing	Processing	Manufacturing	Fraction of
	Processing	Manufacturing	scrap	scrap	downcycled	downcycled	exported final
Product categories	yield λ_P	yield $\lambda_{-}M$	recovery ξ_P	recovery ξ_M	scrap ω_P	$scrap[\omega_M]$	products ψ_P
portable batteries	0.000	0.000	0.000	0.000	0.000	0.000	0.000
mobility batteries	0.850	1.000	1.000	1.000	0.000	0.000	0.000
hydroprocessing catalysts coke	0.970	0.960	1.000	1.000	0.000	0.000	0.000
hydroprocessing catalysts poisoning	0.970	0.960	1.000	1.000	0.000	0.000	0.000
hydroformylation catalysts	0.970	0.010	1.000	1.000	0.000	0.000	0.000
pet precursors catalysts	0.970	0.970	1.000	1.000	0.000	0.000	0.000
dissipative uses	0.970	0.950	1.000	1.000	0.000	0.000	0.380
hard metals	0.980	1.000	1.000	1.000	0.000	0.000	0.460
magnets	0.940	0.960	0.000	0.000	0.500	0.000	0.620
other metallic uses	0.750	0.790	0.680	0.690	0.290	0.450	0.440
superalloys	0.710	0.470	0.950	0.470	0.560	0.890	0.610
		Table /	A.13: Inputs produ	ction			

B

Uncertainty scoring of Input uncertainty scoring

Scores marked with a star (*) base on the evaluation of assessment of Godoy León and Dewulf (2020) as follows:

- Uncertainty score 1: "Very high data quality"
- Uncertainty score 2: "High data quality"
- Uncertainty score 3: "Low data quality"
- Uncertainty score 4: "No data reported"

All other scores base on the perception of the author. The reasoning is explained in Section 2.4.2 and in the captions of the tables.

B.1. Use phase

Product categories	Initial inflow split $arepsilon$	Weibull shape β	Weibull scale T
portable batteries	2*	0*	0*
mobility batteries	2*	1*	4*
hydroprocessing catalysts coke	2*	2*	4*
hydroprocessing catalysts poisoning	2*	2*	4*
hydroformylation catalysts	2*	2*	4*
pet precursors catalysts	2*	1*	4*
dissipative uses	2*	2*	3*
hard metals	2*	1*	2*
magnets	2*	1*	2*
other metallic uses	2*	1*	2*
superalloys	2*	1*	2*

Table B.1: Input uncertainty scoring initial flow split and Weibull parameters for all product categories except for portable batteries

Products	Hoarding rate	Hoarding time (years)
portable batteries	0	0
mobility batteries	2*	2*
hydroprocessing catalysts coke	4*	4*
hydroprocessing catalysts poisoning	4*	4*
hydroformylation catalysts	4*	4*
pet precursors catalysts	4*	4*
dissipative uses	2*	2*
hard metals	3*	2*
magnets	4*	4*
other metallic uses	4*	4*
superalloys	3*	3*

Table B.2: Input uncertainty scoring for hibernating stock of all product categories except for portable batteries

Products	Split portable batteries to products δ
smartphones	4
mobile phones	4
tablets	4
laptops	4
e-bikes	4
power tools	4
others	4

 Table B.3: Input uncertainty scoring split portable batteries to products. The uncertainty is high. The split bases on a single source and was adapted to fit the considered products. The split of portable batteries to products bases on one non-scientific source and is therefore scored with 4.

Products	Weibull scale T	Weibull shape β	to use α	to storage β
smartphones	2 (a)	1 (a)	3 (b)	3 (b)
mobile phones	2 (a)	1 (a)	3 (b)	3 (b)
tablets	2 (a)	1 (a)	3 (b)	3 (b)
laptops	2 (a)	1 (a)	3 (b)	3 (b)
e-bikes	4	2	4	4
power tools	4	2	4	4
others	4	2	4	4

Table B.4: Input uncertainty scoring reuse first life-cycle. (a) the data bases on empirical studies, (b) the data bases on the literature review and assumptions of Glöser-Chahoud et al. (2019), and the rest of the data bases on own assumptions.

Products	Weibull scale T	Weibull shape β	to use γ
smartphones	2 (a)	1 (a)	3 (b)
mobile phones	2 (a)	1 (a)	3 (b)
tablets	2 (a)	1 (a)	3 (b)
laptops	2 (a)	1 (a)	3 (b)
e-bikes	4	2	4
power tools	4	2	4
others	4	2	4

 Table B.5: Input uncertainty scoring hibernating stock after first use-cycle. (a) the data bases on empirical studies, (b) the data bases on the literature review and assumptions of Glöser-Chahoud et al. (2019), and the rest of the data bases on own assumptions.

Products	Weibull scale T	Weibull shape β	to use α	to storage β
smartphones	3	2	4	4
mobile phones	3	2	4	4
tablets	3	2	4	4
laptops	3	2	4	4
e-bikes	5	2	5	5
power tools	5	2	5	5
others	5	2	5	5

 Table B.6: Input uncertainty scoring reuse second life-cycle. The data sources are the same as in Table B.4. It is assumed that the certainty of the data gets worse in the second use cycle (also see Section 2.4.2).

Products	Weibull scale T	Weibull shape β	to use γ
smartphones	3	2	4
mobile phones	3	2	4
tablets	3	2	4
laptops	3	2	4
e-bikes	5	2	5
power tools	5	2	5
others	5	2	5

 Table B.7: Input uncertainty scoring hibernating stock after second use-cycle. The data sources are the same as in Table B.5. It is assumed that the certainty of the data gets worse in the second use cycle (also see Section 2.4.2).

Products	Weibull scale T	Weibull shape β	to storage β
smartphones	5	2	5
mobile phones	5	2	5
tablets	5	2	5
laptops	5	2	5
e-bikes	5	2	5
power tools	5	2	5
others	5	2	5

 Table B.8: Input uncertainty scoring reuse third life-cycle. All inputs have the worst score possible. It is assumed that the base of knowledge of the previous whereabouts of the product is very low.

Products	Weibull scale T	Weibull shape β
smartphones	5	2
mobile phones	5	2
tablets	5	2
laptops	5	2
e-bikes	5	2
power tools	5	2
others	5	2

 Table B.9: Input uncertainty scoring hibernating stock after third use-cycle. It is assumed that the base of knowledge of the previous whereabouts of the product is very low.

B.2. End-of-life phase

Product categories	Fraction of collected EOL products exported ψ_{EOL}	Collection to recycling rate σ	Pre-treatment efficiency λ_{PT}
portable batteries	4*	2*	2*
mobility batteries	4*	4*	4*
hydroprocessing catalysts coke	4*	4*	4*
hydroprocessing catalysts poisoning	4*	4*	4*
hydroformylation catalysts	4*	4*	4*
pet precursors catalysts	4*	4*	4*
dissipative uses	4*	0*	0*
hard metals	4*	3*	1*
magnets	4*	2*	3*
other metallic uses	4*	2*	4*
superalloys	4*	2*	4*

Table B.10: Input uncertainty scoring collection and pre-treatment

Product categories	Chemical	Zn	Downcycling
portable batteries	4*	4*	4*
mobility batteries	4*	4*	4*
hydroprocessing catalysts coke	3*	3*	3*
hydroprocessing catalysts poisoning	3*	3*	3*
hydroformylation catalysts	4*	4*	4*
pet precursors catalysts	4*	4*	4*
dissipative uses	0*	0*	0*
hard metals	3*	3*	3*
magnets	0*	0*	0*
other metallic uses	0*	0*	0*
superalloys	2*	2*	2*
Recycling efficiency	2*	2*	0*

Table B.11: Input uncertainty scoring distribution to recycling processes B and recycling efficiency

Product categories	Co metal or compound	W-Co powder
portable batteries	0*	0*
mobility batteries	4*	4*
hydroprocessing catalysts coke	4*	4*
hydroprocessing catalysts poisoning	4*	4*
hydroformylation catalysts	4*	4*
pet precursors catalysts	4*	4*
dissipative uses	4*	4*
hard metals	4*	4*
magnets	4*	4*
other metallic uses	4*	4*
superalloys	4*	4*
Fraction of exported secondary material \$\psi_{R}	3*	4*

Table B.12: Input uncertainty scoring distribution of secondary material and fraction of exported secondary material

B.3. Production

	Processing	Manufacturing	Processing scrap	Manufacturing scrap	Processing downcycled	Manufacturing downcycled	Fraction of exported final
Product categories	yield λ_P	yield λ_M	recovery ξ_P	recovery ξ_M	scrap ω_P	$scrap[\omega_M]$	products ψ_P
portable batteries	•0	*0	•0	*0	*0	*0	*0
mobility batteries	5 *	4*	4*	4*	4*	4*	4*
hydroprocessing catalysts coke	*ო	*°	4*	4*	4*	4	4*
hydroprocessing catalysts poisoning	* ۳	*°	4*	4*	4*	4*	4*
hydroformylation catalysts	* ۳	4*	4*	4*	4*	*0	4*
pet precursors catalysts	*ო	4*	4*	4*	4*	4*	4*
dissipative uses	* സ	*°	4*	o*	4*	°*	4*
hard metals	∕3*	2*	4*	2*	4*	2*	4*
magnets	∕3*	2*	4*	2*	2*	2*	4*
other metallic uses	*ო	*ෆ	*°	*თ	*സ	*സ	4*
superalloys	2*	2*	3* 3*	2*	3*	2*	4*

Table B.13: Input uncertainty scoring production
Monte Carlo simulations input distributions

This appendix entails three graphs showing how those selected inputs were varied for the Monte Carlo simulations.



Figure C.1: Histogram (a) and defined distribution (b) of the transfer coefficient to use in the first use cycle for e-bikes. The actual input value is 0.07 and the assigned uncertainty score is 4. As it can be seen, the histogram does not really match the defined distribution. This is because the created value was normalized with other transfer coefficients to ensure that they sum up to 1. This is done to insure mass conservation.



Figure C.2: Histogram of the Weibull shape for the in-use time of hard metals. The actual input value is 1.16 and the uncertainty score is 1. As it can be seen, there are no values below 1 (for explanation see Section 2.4.2).



Figure C.3: Histogram (a) and defined distribution (b) of the efficiency of the production process for dissipative uses. The actual input value is 0.97 and the assigned uncertainty score is 3. As it can be seen, the histogram follows the defined distribution (in contrast to Figure C.1). This is because efficiencies do not have to be normalized with any other values.

D

Model application manual and description of digital appendix

The model application manual can be found from the next page onwards. It consist out of a Jupyter Notebook explaining how to use the model and how to use the provided scripts to execute Monte Carlo simulations. The notebook is also provided as html file which can be opened in the browser. The names of those files are:

- model_application_manual.ipynb
- model_application_manual.html

The digital appendix consists of multiple files and folders. The model itself is defined by the following files:

- base_matrace_model.py
- base_reuse_model.py
- combined_reuse_matrace_model.py

The following files are needed to execute and evaluate the Monte Carlo simulations:

- functions.py
- monte_carlo_simulations.py
- monte_carlo_evaluation.py
- monte_carlo_uncertainty.py

Furthermore, there is a notebook (also as html) with all graphs used throughout this thesis:

- used_graphs.ipynb
- used_graphs.html

There are three folders in the digital appendix. The folder "data_cobalt_case_study" holds the data used for the case study as well as the results of the Monte Carlo simulations. The folder "data_model" holds exemplary data used in the model manual. The folder "monte_carlo_results" is necessary in order to execute the Jupyter Notebook containing the model application manual.

Model application manual

This notebook describes how to apply the model on different data as presented in the thesis. It will be explained:

- How to create a data set readable by the model
- How to execute the model itself
- And how to conduct Monte Carlo Simulations with the provided code

How to create a readable data set

Creation of a readable data set

The created model requires an excel file in a specific format. A function was created to create a plain excel file in the right format. The function requires a list entailing the product categories which are treated by the model, a string naming the product category which flows into the reuse part of the model, a list indication the products treated in the reuse part of the model, and the number of considered use cycles in the reuse part. Furthermore, a file name has to be provided. The created file will appear in the "data_model" folder.

```
In [ ]:
```

```
from functions import input_file_creator
```

```
input_file_creator(
    product_categories=['electronics', 'cars', 'industrial applications'],
    category_for_reuse='electronics', products=['shaver', 'phone', 'TV'],
    considered_use_cycles=2, file_name = 'example_file_creator')
```

All sheets of the file are empty and to be filled by the user.

```
In [ ]: import pandas as pd
    pd.read_excel('data_model/example_file_creator.xlsx')
```

Out[]: Product categories share

0	electronics	NaN
1	cars	NaN
2	industrial applications	NaN

```
In [ ]:
```

```
import os
```

os.remove('data_model/example_file_creator.xlsx')

To showcase how to input data, an example file will be considered.

Things to consider populating the data frame

A few sheets are presented as examples. In general, common-sense mistakes need to be avoided.

Ensuring mass conservation when splitting the material flow

The following output shows the first sheet, the split of the initial inflow.

In []:

pd.read_excel('data_model/data_example.xlsx', sheet_name='MaTrace_initial_inflow')

Dut[]:		Product categories	share
	0	electronics	0.412
	1	mobility batteries	0.012
	2	hydroprocessing catalysts coke	0.020
	3	hydroprocessing catalysts poisoning	0.000
	4	hydroformylation catalysts	0.003
	5	pet precursors catalysts	0.017
	6	dissipative uses	0.090
	7	hard metals	0.230
	8	magnets	0.006
	9	other metallic uses	0.210

The sum of this column has to equal 1.

```
Out[]: 1.0
```

The same holds true for the sheet "Reuse_inflow_split".

```
In [ ]:
```

pd.read_excel('data_model/data_example.xlsx', sheet_name='MaTrace_initial_inflow')

Out[]:		Product categories	share
	0	electronics	0.412
	1	mobility batteries	0.012
	2	hydroprocessing catalysts coke	0.020
	3	hydroprocessing catalysts poisoning	0.000
	4	hydroformylation catalysts	0.003
	5	pet precursors catalysts	0.017
	6	dissipative uses	0.090
	7	hard metals	0.230
	8	magnets	0.006
	9	other metallic uses	0.210

And as well for the sheet "MaTrace_D_secondary_material". In this case, the last two rows need to be excluded since the indicate the export and to production rate.

_		-	-	
()	1111			•
0	лu			

Product categories Co metal or compound W-Co powder

0	electronics	0.0000	0
1	mobility batteries	0.0210	0
2	hydroprocessing catalysts coke	0.0135	0
3	hydroprocessing catalysts poisoning	0.0135	0
4	hydroformylation catalysts	0.0050	0
5	pet precursors catalysts	0.0240	0
6	dissipative uses	0.1670	0
7	hard metals	0.3130	1
8	magnets	0.0170	0
9	other metallic uses	0.4260	0
10	export rate	0.1330	0
11	to production rate	0.8670	1

In []:

print(pd.read_excel('data_model/data_example.xlsx', sheet_name='MaTrace_D_secondary_material').iloc[:-2, :]) print() print('Sum of columns without export rate and to production rate:') print(pd.read_excel('data_model/data_example.xlsx', sheet_name='MaTrace_D_secondary_material').iloc[:-2, :].sum(axis=0))

	Product categories	Co metal or compound	W-Co powder
0	electronics	0.0000	0
1	mobility batteries	0.0210	0
2	hydroprocessing catalysts coke	0.0135	0
3	hydroprocessing catalysts poisoning	0.0135	0
4	hydroformylation catalysts	0.0050	0
5	pet precursors catalysts	0.0240	0
6	dissipative uses	0.1670	0
7	hard metals	0.3130	1
8	magnets	0.0170	0
9	other metallic uses	0.4260	0

Sum of columns without export rate and to production rate: Product categories electronicsmobility batterieshydroprocessing c... Co metal or compound 1.0 W-Co powder 1 dtype: object

Furthermore, some transfer coefficients must sum up to 1. In the model one transfer coefficient is sufficient to define a flow which splits into two. However, for the Monte Carlo simulations both are needed.

The following cell shows the example of the sheet "MaTrace_end_of_life":

In []:

pd.read_excel('data_model/data_example.xlsx', sheet_name='MaTrace_end_of_life')

	Product categories	fraction export eol products	fraction collected eol products	collection to recycling rate	postconsumer disposal rate	pre- treatment efficiency
0	electronics	0.2	0.8	0.64	0.36	0.95
1	mobility batteries	0.0	1.0	1.00	0.00	0.85
2	hydroprocessing catalysts coke	0.0	1.0	1.00	0.00	0.85
3	hydroprocessing catalysts poisoning	0.0	1.0	0.00	1.00	0.00
4	hydroformylation catalysts	0.0	1.0	0.90	0.10	0.85
5	pet precursors catalysts	0.0	1.0	0.50	0.50	0.85
6	dissipative uses	0.0	1.0	0.00	1.00	0.00
7	hard metals	0.0	1.0	0.47	0.53	0.77
8	magnets	0.0	1.0	0.99	0.01	0.65
9	other metallic uses	0.0	1.0	0.84	0.16	0.65

The column "fraction export eol products" and "fraction collected eol products" as well as the columns "collection to recycling rate" and "postconsumer disposal" have to sum up to 1 for every product category.

Please also consult the system diagram or the excel file data_example.xlsx to find the respective transfer coefficients.

Defining survival curves

The sheet "MaTrace_in_use_stock" show examples on how to define survival curves in the model implementation:

In []: pd.read_excel('data_model/data_example.xlsx', sheet_name='MaTrace_in_use_stock')

	Product categories	distribution	location	scale	shape
0	electronics	normal	1	4.051716	2.09
1	mobility batteries	weibull	0	10.631501	2.20
2	hydroprocessing catalysts coke	gamma	0	2.894742	2.50
3	hydroprocessing catalysts poisoning	defined_distribution_example_dist	0	0.000000	0.00
4	hydroformylation catalysts	lognormal	1	2.315793	2.50
5	pet precursors catalysts	normal	0	0.578948	2.50
6	dissipative uses	weibull	0	14.435166	3.50
7	hard metals	gamma	0	8.915235	1.16
8	magnets	gompertz	2	13.905022	1.93

Out[

	Product categories	distribution	location	scale	shape
9	other metallic uses	lognormal	0	14.628064	1.47

One can select a distribution via the column "distribution". The normal, lognormal, weibull, gamma, and gompertz distribution are preimplemented. The location, scale and shape factors can be set via the corresponding columns. It is also possible to define own distributions.

An example for this is the distribution for the product category "hydroprocessing catalysts poisoning". The string says "defined_distribution_example_dist". This works in the following way. A distribution is defined in the file "defined_distributions.xlsx" (see next cell).

In []:	pd.read_exc	el('data_model/
Out[]:	example_dis	t example_dist_2
	0 1.0	0 1.00
	1 0.6	0 0.70
	2 0.5	5 0.55
	3 0.5	0 0.50
	4 0.4	5 0.45
	5 0.4	0 0.40
	6 0.3	5 0.35
	7 0.3	0 0.30
	8 0.2	5 0.20
	9 0.2	0 0.20

The file holds survival curves defined by the user. Only the first 10 lines are shown. It is important that the distribution is defined for a sufficient number of years, meaning at least the number of considered years.

To use this distribution in the model, one has to fill the column "distribution" with "defined_distribution_column_name". Hence, when the entry in the column "distribution" says "defined_distribution_example_dist", the defined distribution "example_dist" will be used for this product category. If it says "defined_distribution_exmaple_dist_2" the distribution "example_dist_2" will be used for the product category.

How to execute the model iteself

The first step to execute the model is to load the required data. The model receives a dictionary as input which contains the sheets of the mentioned excel file as pandas data frames. The following code cell creates this dictionary.

```
In [ ]: file_path = 'data_model/data_example.xlsx'
```

data_sheets = pd.ExcelFile(file_path).sheet_names

```
data_dic = {}
for data_sheet in data_sheets:
    try:
        data_dic[data_sheet] = pd.read_excel(file_path,
            sheet_name=data_sheet).set_index('Product categories')
    except:
        data_dic[data_sheet] = pd.read_excel(file_path,
            sheet_name=data_sheet).set_index('Products')
```

Furthermore, the number of years and the start year have to be defined. A pandas dataframe containing the "defined_distributions.xlsx" file has to be passed as well.

In []:

```
number_of_years = 25
start_year = 2022
defined_distributions_pd = pd.read_excel('data_model/defined_distributions.xlsx')
```

Optionally, one can decide to print the state to have an indication whether the model is stuck or how long it will still run. One can also define whether the simplified model output shall differentiate between use cycles. Lastly, one can select the number of considered use cycles (default is 3).

The following code shows the execution of the model:

In []: from combined_reuse_matrace_model import evaluate_cohort_combined_model
matrace_data_dic, reuse_data_dic, graph_data_pd = \
 evaluate_cohort_combined_model(data_dic=data_dic,
 n_years=number_of_years, start_year=start_year,
 defined_distributions_pd= defined_distributions_pd,
 print_state=True, separate_reuse_graph=True, considered_use_cycles=3)

Year 1 of 25 Year 2 of 25 Year 3 of 25 Year 4 of 25 Year 5 of 25 Year 6 of 25 Year 7 of 25 Year 8 of 25 Year 9 of 25 Year 10 of 25 Year 11 of 25 Year 12 of 25 Year 13 of 25 Year 14 of 25 Year 15 of 25 Year 16 of 25 Year 17 of 25 Year 18 of 25 Year 19 of 25 Year 20 of 25 Year 21 of 25 Year 22 of 25 Year 23 of 25 Year 24 of 25 Year 25 of 25 The returned dictionaries "matrace_data_dic" and "reuse_data_dic" are structured in the same way. The first key takes a string containing the considered year. The second key takes a string indicating a stock or a flow in the year. This will then return a pandas containing the values over product categories or products.

The following code cell shows the second keys of "matrace_data_dic".

```
In [ ]: matrace_data_dic['0'].keys()
```

```
Out[]: Index(['U.A use stock', 'U.2 use outflow', 'U.3 hoarding inflow',
 'U.4 no hoarding flow', 'U.B hoarding stock', 'U.5 hoarding outflow',
 'U.6 eol products', 'E.2 exported eol products',
 'E.1 to waste treatment', 'E.3 to pretreatment',
 'E.4 E.5 non-selective collection', 'E.6 to recycling',
 'E.7 pretreatment waste', 'E.12 downcycling', 'recycled w-co powder',
 'co metal compound', 'E.8 recycling waste',
 'E.11 exported recycled materials', 'P.1 total recycled products',
 'P.8 export recycled products', 'U.1 product inflow',
 'P.7 processing waste', 'P.5p downcycled scrap', 'P.4p disposed scrap',
 'P.4m disposed scrap', 'P.4 disposed scrap', 'P.5 downcycled scrap'],
 dtype='object')
```

And the following code cell shows the second keys of "reuse_data_dic".

```
In [ ]:
reuse_data_dic['0'].keys()
Out[ ]:
    Index(['total_use_stock', 'total_hoarding_stock', 'to_disposal_flow',
        'use_stock_1', 'storage_stock_1', 'use_1_to_storage_1_flow',
        'use_1_to_disposal_flow', 'storage_1_to_use_2_flow', 'use_stock_2',
        'use_1_to_use_2_flow', 'storage_1_to_use_2_flow', 'use_stock_2',
        'storage_stock_2', 'use_2_to_storage_2_flow', 'use_2_to_disposal_flow',
        'storage_2_to_disposal_flow', 'use_stock_3', 'storage_stock_3',
        'storage_2_to_use_3_flow', 'use_stock_3', 'storage_stock_3',
        'use_3_to_storage_3_flow', 'use_3_to_disposal_flow',
        'storage_3_to_disposal_flow'],
        dtype='object')
```

The following code cell shows the content.

In []:	<pre>reuse_data_dic['0']['total_use_stock']</pre>					
∩u+[]•	Products					
out[].	computer	0.026567				
	phone	0.045320				
	fan	0.082400				
	dish washer	0.185400				
	e-bikes	0.024720				
	power tools	0.019502				
	Name: total_u	se_stock, dtype: float64				
In []:	<pre>matrace_data_dic['0']['U.A use stock']</pre>					
0+[]].	Products					
out[]:	electronics		0.38391			
	mobility batte	eries	0.012			
	hydroprocessi	ng catalysts coke	0.02			
	hydroprocessi	ng catalysts poisoning	0.0			
	hydroformylat	ion catalysts	0.003			
	pet precursor	s catalysts	0.0085			

dissipative uses	0.09
hard metals	0.23
magnets	0.006
other metallic uses	0.21
Name: U.A use stock, dtype: object	

The retuned pandas dataframe "graph_data_pd" contains the stocks and the accumulated outflows.

In []: graph_data_pd

<

Out[]:		Electronics 1st use	Electronics 2nd use	Electronics 3rd use	Mobility batteries	Hydroprocessing catalysts coke	Hydroprocessing catalysts poisoning	Hydroform c
	Year							
	2022	0.377540	0.005548	0.000821	0.012176	0.020113	0.000113	0
	2023	0.281169	0.017890	0.001593	0.012589	0.020090	0.000421	0
	2024	0.199773	0.026279	0.002235	0.012901	0.019297	0.000728	0
	2025	0.142881	0.031148	0.002443	0.012958	0.017849	0.000909	0
	2026	0.106036	0.032665	0.002545	0.012749	0.016026	0.001010	0
	2027	0.083086	0.031205	0.002917	0.012286	0.014056	0.001063	0
	2028	0.068224	0.028201	0.003453	0.011605	0.012104	0.001080	0
	2029	0.057489	0.024919	0.003829	0.010758	0.010275	0.001073	0
	2030	0.047519	0.021968	0.003884	0.009806	0.008634	0.001056	0
	2031	0.041138	0.019131	0.003691	0.008783	0.007190	0.001019	0
	2032	0.034343	0.016645	0.003386	0.007755	0.005957	0.000974	0
	2033	0.028269	0.014450	0.003054	0.006757	0.004920	0.000924	0
	2034	0.022851	0.012621	0.002727	0.005817	0.004056	0.000861	0
	2035	0.018026	0.011123	0.002423	0.004960	0.003344	0.000788	0
	2036	0.014967	0.009819	0.002153	0.004191	0.002755	0.000703	0
	2037	0.012373	0.008728	0.001925	0.003523	0.002274	0.000615	0
	2038	0.010187	0.007783	0.001740	0.002954	0.001882	0.000538	0
	2039	0.008353	0.006950	0.001594	0.002477	0.001563	0.000472	0
	2040	0.006823	0.006196	0.001475	0.002081	0.001302	0.000416	0
	2041	0.005552	0.005507	0.001374	0.001758	0.001089	0.000368	0
	2042	0.004501	0.004883	0.001281	0.001493	0.000916	0.000327	0
	2043	0.003637	0.004319	0.001191	0.001278	0.000773	0.000291	0
	2044	0.002928	0.003814	0.001101	0.001101	0.000656	0.000258	0
	2045	0.002350	0.003365	0.001011	0.000955	0.000559	0.000230	0
	2046	0.001880	0.002968	0.000923	0.000834	0.000479	0.000204	0

It is easily possible to create a stacked area chart out of it.



How to conduct Monte Carlo simulations with the provided code

2035 Year

The code to conduct Monte Carlo simulations consists of three files which need to be executed one after another:

2040

2045

monte_carlo_simulations.py

2025

2030

0.0

- monte_carlo_evaluation.py
- monte_carlo_uncertainty.py

In the following it will be discussed what each file does and how to use it. To do so, files and the data used in the cobalt case study will be used. Each of the files has the variable "proof_of_concept" in the very top. If this is "True", an exemplary Monte Carlo simulation and evaluation considering 10 runs will be conducted. If it is set to "False" the results as presented in the thesis will be reproduced.

```
In [ ]: # The folder holding the results of the following demonstration
# needs to be deleted in order to be created and populated.
# This is a necessary step but not relevant in the context of
# the explenation.
import shutil
try:
    shutil.rmtree('monte_carlo_results/proof_of_concept/')
except:
    print('Dictionary does not excist.')
```

Executing "monte_carlo_simulations.py"

At the start of the file, the user can adjust the settings. Those are the number of runs ("n_runs"), the number of the considered years ("n_years"), the start year ("start_year"), and the number of the considered use cycles ("considered_use_cycles").

Then, the data to be loaded needs to be specified. Firstly, the file entailing the survival curves designed by the user has to be loaded (see above). Secondly, the path to the input data (format as described above) has to be specified. Lastly, the excel file containing the uncertainty rating has to be set.

This file must entail the same sheet names as the input file. Instead of the data, the colums must hold the uncertainty score (0 to 5, as explained in the main body of the thesis).

If an input shall not be varied, the column can be either deleted from the file holding the uncertainty scores, or all values can be set to 0. Only numerical inputs can be considered.

The following cell shows one sheet of the excel holding the input data. The next one shows the uncertainty ratings of those inputs.

In []:

pd.read_excel('data_cobalt_case_study/data_input_cobalt_extended_data_set.xlsx', sh

Out[]:		Product categories	distribution	location	scale	shape
	0	portable batteries	weibull	0	4.051716	2.09
	1	mobility batteries	weibull	0	10.631501	2.20
	2	hydroprocessing catalysts coke	weibull	0	2.894742	2.50
	3	hydroprocessing catalysts poisoning	weibull	0	1.505266	2.50
	4	hydroformylation catalysts	weibull	0	2.315793	2.50
	5	pet precursors catalysts	weibull	0	0.578948	2.50
	6	dissipative uses	weibull	0	14.435166	3.50
	7	hard metals	weibull	0	8.915235	1.16
	8	magnets	weibull	0	13.905022	1.93
	9	other metallic uses	weibull	0	14.628064	1.47
	10	superalloys	weibull	0	17.282543	1.74

^{&#}x27;data_cobalt_case_study/data_uncertainty_rating_cobalt_case_study.xlsx',
sheet_name='MaTrace_in_use_stock')

	Product categories	shape	scale
0	portable batteries	0	0
1	mobility batteries	1	4
2	hydroprocessing catalysts coke	2	4
3	hydroprocessing catalysts poisoning	2	4
4	hydroformylation catalysts	2	4
5	pet precursors catalysts	1	4
6	dissipative uses	2	3
7	hard metals	1	2
8	magnets	1	2
9	other metallic uses	1	2
10	superalloys	1	2

As it can be seen, only the uncertainty scores for the columns "shape" and "scale" appear. This is intended since only numerical values can be varied by this implementation and the location of the Weibull distribution is a factor which was not considered.

Once the paths and the settings are defined, the script can be run. The next cell executes the script for 10 runs.

In []:

Out[]:

%run monte_carlo_simulations.py

Creating dictionary for results. Path: monte_carlo_results/proof_of_concept Importing data Creating inputs for Monte Carlo simulations based on uncertainty score Normalize inputs c:\Users\rapha\Desktop\master_thesis_hand_in\monte_carlo_simulations.py:154: Perfor manceWarning: indexing past lexsort depth may impact performance. for column in input_pd.loc[:, ("MaTrace_initial_inflow", "share")].columns: Starting runs Run 1 of 10 Mass balance run 0: True Run 2 of 10 Mass balance run 1: True Run 3 of 10 Mass balance run 2: True Run 4 of 10 Mass balance run 3: True Run 5 of 10 Mass balance run 4: True Run 6 of 10 Mass balance run 5: True Run 7 of 10 Mass balance run 6: True Run 8 of 10 Mass balance run 7: True Run 9 of 10 Mass balance run 8: True Run 10 of 10 Mass balance run 9: True Total time - seconds: 21, hours: 0.00583333333333333333

Time per run: 2.1 Inputs and results are stored.

The printed output reflects the steps of the code. After the data is imported, random numbers following the defined distributions are created. Since some split vectors and transfer coefficients will not sum up to 1 anymore (see above) they have to be normalized. Afterwards, the experiments are run. The data is collected in a dictionary. If several thousand runs are executed, the dictionary is dumped in splits to decrease the runtime.

The results and a file containing the used model inputs can be found in the folder "monte_carlo_results/proof_of_concept".

The following cell shows part of the saved inputs. The rows represent runs.

```
In [ ]:
```

```
import pickle
with open('monte_carlo_results/proof_of_concept/inputs.pkl', 'rb') as f:
    input_pd = pickle.load(f)
input_pd
```

Out[]: sheet

column

item	portable batteries	mobility batteries	hydroprocessing catalysts coke	hydroprocessing catalysts poisoning	hydroformylation catalysts	pet precursors catalysts	diss
0	0.195626	0.069050	0.049883	0.023999	0.059167	0.004703	0.
1	0.221689	0.031145	0.057574	0.063652	0.049747	0.142862	0.
2	0.203971	0.067055	0.028344	0.034823	0.238051	0.132933	0.
3	0.411710	0.092229	0.039897	0.041033	0.107387	0.019387	0.
4	0.279781	0.026076	0.036750	0.036071	0.015810	0.016235	0.
5	0.282791	0.069646	0.045512	0.080305	0.017813	0.049054	0.
6	0.321172	0.040106	0.077813	0.053932	0.036862	0.058657	0.
7	0.252250	0.032541	0.046556	0.083294	0.096109	0.016466	0.
8	0.267555	0.034854	0.015555	0.001666	0.069714	0.074432	0.
9	0.326831	0.035760	0.133187	0.041984	0.067035	0.004005	0.

10 rows × 541 columns

<	>

The results are saved in the form of nested dictionaries:

In []:

with open('monte_carlo_results/proof_of_concept/results.pkl', 'rb') as f: results_dic = pickle.load(f)

The fist key represents runs:

```
In [ ]:
```

```
results_dic.keys()
```

The second part specifies from which part of the model the data is coming from:

In []: results_dic['0'].keys()

```
Out[ ]: dict_keys(['matrace_data_dic', 'reuse_data_dic'])
```

Since this data structure is hard to handle, the script "monte_carlo_evaluation.py" serves the purpose to transfer the data into multidimensional pandas data frames.

Executing "monte_carlo_evaluation.py"

In order to execute this script, one has to adjust the settings in the beginning of the file so they are the same as the used ones in "mone_carlo_simulations.py". The path to the results has to be defined. Then the file is ready to be executed.

In []:

%run monte_carlo_evaluation.py

```
Loading file:
Create dictionary:
Start treating data
run 1 of 10
run 2 of 10
run 3 of 10
run 4 of 10
run 5 of 10
run 6 of 10
run 7 of 10
run 8 of 10
run 9 of 10
run 10 of 10
Compact results are stored.
```

This script creates the multidimensional pandas data frame 'compact_results'. The first key entails the runs, the second the stock or flow, and the third one the product or the product category.

Out[]: run

```
stock_flow
```

```
total_use_stock 1
```

item	sum	smartphones	mobile phones	tablets	laptops	e-bikes	power tools	others	
2015	0.195626	0.057242	0.047268	0.02614	0.048304	0.001794	0.006937	0.007942	
2016	0.177952	0.053884	0.036155	0.02468	0.047307	0.001774	0.006729	0.007423	0.01
2017	0.1475	0.046705	0.021888	0.020836	0.044188	0.001634	0.006348	0.005901	0.02
2018	0.116835	0.037369	0.012804	0.015898	0.039273	0.001304	0.005903	0.004284	0.02
2019	0.090162	0.027998	0.008145	0.011246	0.033273	0.000864	0.005448	0.003187	0.02

.

stock_flow

item	sum	smartphones	mobile phones	tablets	laptops	e-bikes	power tools	others	
2020	0.069089	0.020143	0.006213	0.007728	0.027022	0.000502	0.005007	0.002473	0.02
2021	0.053197	0.01421	0.00547	0.005481	0.021222	0.000322	0.00459	0.001902	0.02
2022	0.041158	0.009933	0.004847	0.004179	0.016291	0.000263	0.004195	0.00145	0.03
2023	0.031819	0.006883	0.003998	0.003389	0.012354	0.000232	0.003823	0.001141	0.03
2024	0.024471	0.004703	0.003038	0.002799	0.009323	0.000194	0.003472	0.000942	0.02
2025	0.01869	0.003151	0.002168	0.002264	0.007016	0.00015	0.003139	0.000802	0.02
2026	0.014169	0.002059	0.001487	0.001757	0.005248	0.000108	0.002824	0.000686	0.01
2027	0.010659	0.00131	0.000997	0.001299	0.003875	0.000073	0.002528	0.000577	0.01
2028	0.007954	0.000811	0.000655	0.000915	0.002804	0.000046	0.00225	0.000472	0.00
2029	0.005892	0.000489	0.00042	0.000614	0.001976	0.000027	0.001991	0.000375	0.00
2030	0.004344	0.000287	0.00026	0.000393	0.001349	0.000015	0.001752	0.000288	0.00
2031	0.003204	0.000165	0.000153	0.00024	0.000891	0.000008	0.001532	0.000213	0.00
2032	0.002377	0.000092	0.000086	0.00014	0.000569	0.000004	0.001332	0.000153	0.00
2033	0.001785	0.00005	0.000046	0.000078	0.000351	0.000002	0.001152	0.000106	0.00
2034	0.001363	0.000027	0.000023	0.000042	0.00021	0.000001	0.00099	0.000071	0.00
2035	0.001061	0.000014	0.000011	0.000021	0.000121	0.0	0.000847	0.000046	0.00
2036	0.00084	0.000007	0.000005	0.00001	0.000068	0.0	0.000721	0.000029	0.00
2037	0.000675	0.000004	0.000002	0.000005	0.000037	0.0	0.00061	0.000018	0.00
2038	0.000549	0.000002	0.000001	0.000002	0.000019	0.0	0.000515	0.00001	0.00
2039	0.00045	0.000001	0.0	0.000001	0.00001	0.0	0.000432	0.000006	0.00
2040	0.000371	0.0	0.0	0.0	0.000005	0.0	0.000361	0.000003	0.00
2041	0.000306	0.0	0.0	0.0	0.000002	0.0	0.000301	0.000002	0.00
2042	0.000252	0.0	0.0	0.0	0.000001	0.0	0.00025	0.000001	0.00
2043	0.000208	0.0	0.0	0.0	0.000001	0.0	0.000207	0.000001	0.00
2044	0.000171	0.0	0.0	0.0	0.0	0.0	0.000171	0.0	0.00
2045	0.000141	0.0	0.0	0.0	0.0	0.0	0.00014	0.0	0.00
2046	0.000115	0.0	0.0	0.0	0.0	0.0	0.000115	0.0	0.00
2047	0.000094	0.0	0.0	0.0	0.0	0.0	0.000094	0.0	0.00
2048	0.000077	0.0	0.0	0.0	0.0	0.0	0.000077	0.0	0.00
2049	0.000063	0.0	0.0	0.0	0.0	0.0	0.000063	0.0	0.00
2050	0.000051	0.0	0.0	0.0	0.0	0.0	0.000051	0.0	0.00

36 rows × 5240 columns

.

Executing "monte_carlo_uncertainty.py"

In order to execute this script, one has to adjust the settings in the beginning of the file so they are the same as the used ones in "mone_carlo_simulations.py" and "monte_carlo_evaluation.py". The path to the results has to be defined. Then the file is ready to be executed.

The file calculates for all considered years the spearman correlation and the normalized spearman square correlation between all inputs and the total in-use stock, the total hibernating stock, the total disposal flow, and the total export flow.

The results are written into the files "monte_carlo_results" (results of mentioned stocks and flows over years and runs), "speaman_results_abs" (spearman correlation between inputs and outputs over years) and "spearman_results_normalized" (normalized spareman square correlation between inputs and outputs over years).

```
In [ ]:
```

```
%run monte_carlo_uncertainty.py
```

```
Calculate spearman correlation between inputs and:
['total_use_stock', 'total_hoarding_stock', 'to_disposal_flow', 'U.B hoarding stoc
k', 'U.A use stock', 'total_export', 'total_disposal']
Year 1 of 36
c:\Users\rapha\.conda\envs\master_thesis\lib\site-packages\scipy\stats\stats.py:448
4: SpearmanRConstantInputWarning: An input array is constant; the correlation coeff
icient is not defined.
  warnings.warn(SpearmanRConstantInputWarning())
Year 2 of 36
Year 3 of 36
Year 4 of 36
Year 5 of 36
Year 6 of 36
Year 7 of 36
Year 8 of 36
Year 9 of 36
Year 10 of 36
Year 11 of 36
Year 12 of 36
Year 13 of 36
Year 14 of 36
Year 15 of 36
Year 16 of 36
Year 17 of 36
Year 18 of 36
Year 19 of 36
Year 20 of 36
Year 21 of 36
Year 22 of 36
Year 23 of 36
Year 24 of 36
Year 25 of 36
Year 26 of 36
Year 27 of 36
Year 28 of 36
Year 29 of 36
Year 30 of 36
Year 31 of 36
Year 32 of 36
```

2

```
Year 33 of 36
Year 34 of 36
Year 35 of 36
Year 36 of 36
Export Monte Carlo results
Export spearman absolute results
Export spearman normalized results
```

The following cell shows a part of the table displaying the normalized square spearmen correlation.

In []:

```
with open(
    'monte_carlo_results/proof_of_concept/spearman_results_normalized.pkl',
    'rb') as f:
    spearman_results_normalized_pd= pickle.load(f)
spearman_results_normalized_pd
```

Out[]: result_item

sheet

column

item	portable batteries	mobility batteries	hydroprocessing catalysts coke	hydroprocessing catalysts poisoning	hydroformylation catalysts	pet precursors catalysts	(
0	2.045029	0.114251	0.081801	0.138889	0.033126	0.138889	
1	1.986973	0.039561	0.062895	0.081441	0.039561	0.179560	
2	1.887136	0.012615	0.032919	0.021573	0.012615	0.151159	
3	1.887136	0.012615	0.032919	0.021573	0.012615	0.151159	
4	1.887136	0.012615	0.032919	0.021573	0.012615	0.151159	
5	1.949861	0.081918	0.021740	0.039793	0.009102	0.195656	
6	1.949861	0.081918	0.021740	0.039793	0.009102	0.195656	
7	1.956062	0.228274	0.021809	0.055012	0.000679	0.092442	
8	1.812508	0.063447	0.001886	0.000075	0.009129	0.103281	
9	1.887136	0.012615	0.032919	0.021573	0.012615	0.151159	
10	1.769097	0.000663	0.005964	0.008910	0.021281	0.206843	
11	1.643312	0.001850	0.001850	0.003627	0.053960	0.177721	
12	1.331594	0.053264	0.038651	0.003580	0.089504	0.190040	
13	1.331594	0.053264	0.038651	0.003580	0.089504	0.190040	
14	1.305751	0.021333	0.090426	0.021333	0.039049	0.207352	
15	1.049711	0.001853	0.352918	0.101480	0.026760	0.071236	
16	1.049711	0.001853	0.352918	0.101480	0.026760	0.071236	
17	1.049711	0.001853	0.352918	0.101480	0.026760	0.071236	
18	1.200942	0.012583	0.334245	0.091212	0.000670	0.054280	
19	0.971130	0.033540	0.549488	0.182605	0.040232	0.012853	
20	1.022361	0.209790	0.442806	0.062810	0.054445	0.000075	

result_item

sheet

column

per precursors catalysts	hydroformylation catalysts	hydroprocessing catalysts poisoning	hydroprocessing catalysts coke	mobility batteries	portable batteries	item
0.001843	0.021310	0.012461	0.351060	0.292661	1.079578	21
0.001843	0.021310	0.012461	0.351060	0.292661	1.079578	22
0.003694	0.092341	0.016961	0.379993	0.424015	0.962534	23
0.006090	0.102932	0.054812	0.400676	0.569097	0.893307	24
0.006090	0.102932	0.054812	0.400676	0.569097	0.893307	25
0.001845	0.311830	0.000664	0.274632	0.666098	0.752894	26
0.001845	0.311830	0.000664	0.274632	0.666098	0.752894	27
0.026747	0.294067	0.003630	0.136994	0.697121	0.880273	28
0.026747	0.294067	0.003630	0.136994	0.697121	0.880273	29
0.026747	0.294067	0.003630	0.136994	0.697121	0.880273	30
0.026747	0.294067	0.003630	0.136994	0.697121	0.880273	31
0.021238	0.176442	0.016535	0.123532	0.663220	0.779623	32
0.021238	0.176442	0.016535	0.123532	0.663220	0.779623	33
0.021238	0.176442	0.016535	0.123532	0.663220	0.779623	34
0.021238	0.176442	0.016535	0.123532	0.663220	0.779623	35

36 rows × 3787 columns

>

Using this data, one can create plots displaying the contribution of an input to the uncertainty of an output. The following graph is not representative since only 10 runs were conducted.

In []:

```
graph_pd = spearman_results_normalized_pd.groupby(level=[0,1,2], axis = 1).sum()
graph_pd = graph_pd['U.A use stock'].copy()
graph_pd.columns = ['_'.join(col) for col in graph_pd.columns]
graph_pd

dic_legend = {'n_initial_products_Share': r'$\eta$: Initial inflow distribution',
    'n_use_1_in_use_Weibull scale': r'$S_{Uk}$: Weibull scale',
    'n_use_1_in_use_Weibull shape': r'$S_{Uk}$: Weibull shape',
    'reuse_split_split': r'$\delta$: Split portable batteries to products'
    handles_names_list = []

fig, ax = plt.subplots()
stacks = ax.stackplot(graph_pd.index, graph_pd.transpose().to_numpy())

plt.ylabel('Normalized square of Spearman\'s rank correlation in %')
plt.xlabel('Year')
```

